# Can Successful Schools Replicate? Scaling Up Boston's Charter School Sector<sup>†</sup>

By SARAH R. COHODES, ELIZABETH M. SETREN, AND CHRISTOPHER R. WALTERS\*

Can schools that boost student outcomes reproduce their success at new campuses? We study a policy reform that allowed effective charter schools in Boston, Massachusetts to replicate their school models at new locations. Estimates based on randomized admission lotteries show that replication charter schools generate large achievement gains on par with those produced by their parent campuses. The average effectiveness of Boston's charter middle school sector increased after the reform despite a doubling of charter market share. An exploration of mechanisms shows that Boston charter schools compress the distribution of teacher effectiveness and may reduce the returns to teacher experience, suggesting the highly standardized practices in place at charter schools may facilitate replicability. (JEL H75, I21, I28)

The feasibility of scaling up effective programs is a perennial problem in social policy. Successful demonstration projects often fail to reproduce their effects at scale. In the education sphere, for example, recent large-scale studies of early childhood programs, class size reductions, and the Success For All curriculum show effects that fall short of the impressive gains seen in smaller-scale evaluations of similar interventions (Heckman et al. 2010; Heckman, Pinto, and Savelyev 2013; Puma et al. 2012; Krueger 1999; Jepsen and Rivkin 2009; Borman et al. 2007; Quint et al. 2015). This suggests that in some cases the success of programs may be driven by unique inputs or population characteristics such as special teachers, school leaders, peer environments, or other factors that cannot be easily replicated (see Banerjee et al. 2017 on the challenges of scaling up demonstration programs, including

<sup> $\dagger$ </sup>Go to https://doi.org/10.1257/pol.20190259 to visit the article page for additional materials and author disclosure statement(s) or to comment in the online discussion forum.

<sup>\*</sup>Cohodes: Teachers College Columbia University, 525 West 125th Street, New York, NY 10027, and NBER (email: cohodes@tc.columbia.edu); Setren: Tufts University, 8 Upper Campus Road, Medford, MA 02155 (email: Elizabeth.Setren@tufts.edu); Walters: University of California Berkeley, 530 Evans Hall, Berkeley, CA 94720, and NBER (email: crwalters@econ.berkeley.edu). John Friedman was coeditor for this article. Special thanks go to Carrie Conaway, Cliff Chuang, the staff of the Massachusetts Department of Elementary and Secondary Education, and Boston's charter schools for data and assistance. We also thank Josh Angrist, Bob Gibbons, Caroline Hoxby, Parag Pathak, Derek Neal, Eric Taylor, and seminar participants at the NBER Education Program Meetings, Columbia Teachers College Economics of Education Workshop, the Association for Education Finance and Policy Conference, the Society for Research on Educational Effectiveness Conference, Harvard Graduate School of Education, Federal Reserve Bank of New York, MIT Organizational Economics Lunch, MIT Labor Lunch, and University of Michigan Causal Inference for Education Research Seminar for helpful comments. We are grateful to the school leaders who shared their experiences expanding their charter networks: Shane Dunn, Jon Clark, Will Austin, Anna Hall, and Dana Lehman. Setren was supported by a National Science Foundation Graduate Research Fellowship.

general equilibrium and spillover effects; see Davis et al. 2017 on the role of labor supply in scale-up efforts).

The potential for sustained success at scale is of particular interest for "No Excuses" charter schools, a recent educational innovation that has demonstrated promise for low-income urban students. These schools share a set of practices that includes high expectations, strict discipline, increased time in school, frequent teacher feedback, high-intensity tutoring, and data-driven instruction. Evidence based on randomized admission lotteries shows that No Excuses charter schools generate test score gains large enough to close racial and socioeconomic achievement gaps in a short time, as well as improvements in longer-run outcomes like teen pregnancy and four-year college attendance (Abdulkadiroğlu et al. 2011, 2017; Angrist, Pathak, and Walters 2013; Angrist et al. 2012, 2016; Dobbie and Fryer 2011, 2013, 2015; Tuttle et al. 2013; Walters 2018). Other recent studies demonstrate positive effects of No Excuses policies when implemented in traditional public schools or in low-performing schools converted to charter status (Fryer 2014, Abdulkadiroğlu et al. 2016). No school district has adopted these policies on a wide scale, however, and No Excuses charters serve small shares of students in many of the cities where they operate. It therefore remains an open question whether the effects documented in previous research can be replicated at a larger scale. Replicability is also a core issue for charter schools more generally, since by design charters are intended to serve as laboratories of innovation and spread successful educational practices.<sup>1</sup>

We address this question using a recent policy change that expanded the charter school sector in Boston, Massachusetts, a city where most charter schools operate according to No Excuses principles. In 2010, Massachusetts passed a comprehensive education reform law that raised the state's cap on the fraction of funding dedicated to charter school tuition payments in low-performing districts. Charter operators that the state deemed "proven providers" with track records of success were permitted to expand existing campuses or open new schools in these districts. As a result, the number of charter schools in Boston increased from 16 to 32 between 2010 and 2014, with most of these new campuses linked to existing No Excuses charter schools. This expansion led to dramatic growth in charter market share in Boston: the fraction of sixth-grade students attending charter schools increased from 15 to 31 percent between 2010 and 2015.

This increase is equal to the difference in charter attendance rates between the fifteenth-ranked and one hundredth-ranked school districts in the United States (charter share rankings from 2016–2017; see David, Hesla, and Pendergrass 2017).<sup>2</sup> Boston's charter expansion is therefore a large, policy-relevant change in charter share, and it occurs in a single education market. Among charter expansions studied in the previous literature, the closest analog is the growth of the Knowledge is Power Program (KIPP) network of charter schools (Tuttle et al. 2015). Between 2010 and 2015, the KIPP network doubled its student population, from about 27,000

<sup>&</sup>lt;sup>1</sup>Massachusetts charters are required by law to disseminate their "best practices." For details on the Massachusetts policy, see http://www.doe.mass.edu/charter/bestpractices/.

 $<sup>^{2}</sup>$ School district rankings are for all grade levels, and the charter share we focus on here is for middle schools. The leap for all grade levels is a move from approximately 208th to 63rd.

to 55,000 students. Nonexperimental estimates comparing KIPP students to observably similar non-KIPP students showed that the network continued to boost student achievement over this period of expansion, but that these gains were smaller in the years of greater expansion. However, the KIPP expansion differs on important dimensions from the Boston expansion studied here, as the growth of KIPP schools was diffuse, over many cities, rather than concentrated in a particular locality. The policy we study is also distinct from the turnaround strategies studied in Fryer (2014) and Abdulkadiroğlu et al. (2016), which involved transformations of extant schools rather than new entrants to a market.

Other localities face policy choices regarding charter expansions similar to the policy change we study here. New York City reached its cap on the number of charter schools in the city in winter 2019 with 10 percent of students enrolled in charter schools. An increase in the charter cap could result in an influx of charter schools, and would follow similar cap increases in 2007 and 2010 (New York State Department of Education 2019). Massachusetts voters faced a decision about a referendum for another similar charter cap increase in 2016, which did not pass. Boston has again reached the cap on charter schools and thus the state may face future legislation about changing the cap. Several other states have reached their overall caps on charter schools or have limited remaining growth, including Connecticut, Maine, and Rhode Island, setting the stage for policy decisions regarding further growth (Ziebarth and Palmer 2018). The federal government also supports charter school replication, with several charter school networks receiving very large grants to replicate their models, including 2019 awards of over \$100 million to IDEA Public Schools and over \$85 million to KIPP.<sup>3</sup>

We use records from randomized charter school admission lotteries to study changes in the effectiveness of Boston's charter middle school sector during this period of rapid expansion. By comparing the outcomes of students who randomly receive lottery offers to those who do not, we eliminate selection bias that plagues observational comparisons and generate reliable estimates of the causal effects of charter school attendance. The lottery records used here cover 14 of the 15 charter schools admitting students in fifth or sixth grade during the time period of our study. This is important in light of evidence that schools with more readily available lottery records tend to be more effective (Abdulkadiroğlu et al. 2011). Unlike previous studies that focus on subsets of oversubscribed charter schools, our estimates provide a representative picture of the effectiveness of the Boston charter middle school sector before and after expansion.

Consistent with past work, our estimates for cohorts applying before 2010 show large positive impacts of charter attendance on test scores. Specifically, a year of attendance at a Boston charter middle school boosted math achievement by between 0.18 and 0.32 standard deviations ( $\sigma$ ) and increased English Language Arts (ELA) achievement by about 0.1 $\sigma$  during this period. Our results also indicate that policymakers selected more effective schools for expansion: proven providers produced larger effects than other charter schools before the reform.

<sup>&</sup>lt;sup>3</sup>For details on these awards, see https://oese.ed.gov/offices/office-of-discretionary-grants-support-services/ charter-school-programs/state-charter-school-facilities-incentive-grants/awards/.

We make two main contributions to the literature. Our first contribution is to show that a swift, within-market scale-up of a "proven" policy can be successful. Estimates for the post-reform period reveal that Boston's charter sector remained effective while doubling in size. Proven providers and other existing charters maintained their effectiveness after the reform, while expansion charters generate achievement gains comparable to those of their parent schools. Moreover, expansion charters produce these large impacts while enrolling students that appear more representative of the general Boston population than students at other charters. Together, the estimates for new and existing schools imply an increase in overall charter effectiveness despite the substantial growth in charter market share after the 2010 reform. This is the first evidence on the effects of a large scale-up of an effective charter school sector within a single market.

Our second contribution is a detailed investigation of the mechanisms that made this successful within-market expansion possible. The Boston context benefits from a large number of recent college graduates in the teacher labor market, a long track record with charters, and a geographically desirable location, which may limit the generalizability of our findings. However, by exploring which mechanisms make possible successful replication at scale, we provide more general evidence on what contributes to successful schools. We explore the roles of student composition, public school alternatives, and school practices in mediating the effectiveness of expansion charter schools. Though changes in demographic composition contributed modestly to the positive impacts of new charters, neither changes in the student body nor the quality of applicants' fallback traditional public schools explain the pattern of results. Instead, it appears that proven providers successfully transmitted hiring and pedagogical practices to new campuses. An analysis of teacher value-added indicates that charter schools compress the distribution of teacher effectiveness and may reduce returns to experience while also employing a large share of new and inexperienced teachers. These findings are consistent with the possibility that Boston charter schools' use of highly standardized school models that limit teacher discretion may facilitate replicability in new contexts.

The next section provides background on charter schools in Boston and the charter expansion reform. Section II describes the data and Section III details the empirical framework used to analyze it. Section IV presents lottery-based estimates of charter school effects before and after the reform. Section V explores the role of student composition and fallback schools, and Section VI discusses charter management practices and teacher productivity. Section VII offers concluding thoughts.

### I. Background

### A. Charter Schools in Boston

The first charter schools in Boston opened in 1994. Boston charters offer a different educational experience than traditional public schools operating in the Boston Public Schools (BPS) district. Table 1 compares inputs and practices of BPS schools and the 14 charter middle schools in our analysis sample (described in more detail later on). Columns 1 and 5 of panel A show that charter students spend more days

	All charters (1)	Proven providers (2)	Expansion charters (3)	Other charters (4)	Boston Public Schools (5)
Panel A. Comparison with traditional public schools	5				
Days per year	185.9	183.8	186.6	187.3	180.0
Hours per day	8.1	8.1	8.0	8.0	7.3
Percent of teachers licensed in teaching assignment	47.2	45.7	42.8	59.6	95.1
Percent of core academic classes taught by highly qualified teachers	78.7	88.9	68.7	88.4	93.2
Average years of teaching experience in Massachusetts for teachers	2.6	2.9	1.6	3.3	12.3
Student/teacher ratio	11.2	12.5	10.2	11.7	11.7
Average per-pupil expenditure	\$17,041	\$17,900	\$17,831	\$14,052	\$18,766
Title 1 eligible	1.0	1.0	1.0	1.0	1.0
Panel B. Charter school characteristics					
Years open through 2012–2013	7.4	11.0	2.4	14.3	
Tutoring	1.0	1.0	1.0	1.0	
Homework help program	0.4	0.3	0.3	1.0	
Saturday programming	0.6	0.5	0.6	0.7	
School break programming	0.5	0.5	0.3	1.0	
No Excuses index	0.9	0.9	0.9	0.8	
Contact parents at least monthly	0.5	0.5	0.4	0.7	
Distance from parent campus (miles)	—	—	3.1	—	
Observations (schools)	14	4	7	3	5

TABLE 1—SCHOOL CHARACTERISTICS

*Notes:* This table displays characteristics for charter schools in the analysis sample along with Boston Public Schools (BPS) district schools serving middle school grades. Characteristics are measured in the 2012–2013 school year. Per-pupil expenditure is CPI-adjusted to 2015 dollars. The No Excuses index is an equally weighted average of No Excuses characteristics mentioned in charter school annual reports, described in detail in footnote 5.

*Sources:* Data sources include charter school annual reports, school websites, Massachusetts Department of Elementary and Secondary Education (MA DESE) School District Profiles, and MA DESE Education Personnel Information Management System (EPIMS) data.

per year and hours per day in school than BPS students. Charter teachers tend to be younger and less experienced than BPS teachers; as a result, they are much less likely to be licensed or designated highly qualified.<sup>4</sup> BPS and charter schools have similar student/teacher ratios, but charters spend somewhat less money per pupil (\$18,766 versus \$17,041), a difference driven by lower salaries and retirement costs for their less experienced teachers (Setren, forthcoming).

Boston charter schools commonly subscribe to No Excuses pedagogy, an approach that utilizes strict discipline, extended instructional time, selective teacher hiring, frequent testing, high expectations, teacher feedback, data-driven instruction, and tutoring (Carter 2000, Thernstrom and Thernstrom 2004). Panel B of Table 1 reports the mean of an index of No Excuses policies, constructed as an equally weighted average of features typically associated with the No Excuses model.<sup>5</sup> On average, Boston

<sup>4</sup> In the time period of our study, teachers were designated highly qualified if they possessed a Massachusetts teaching license and a bachelor's degree, and passed a state examination or held a degree in their subject area. The highly qualified label was discontinued with the passage of the federal Every Student Succeeds Act (ESSA) in 2015.

<sup>5</sup>The No Excuses index is an average of indicators equal to one if the following items are mentioned in a school's annual report: high expectations for academics, high expectations for behavior, strict behavior code,

charter schools implement 90 percent of these policies. Charters also commonly offer Saturday school and school break programming for homework help and tutoring. These practices differ markedly from practices at BPS schools and at nonurban charter schools in Massachusetts (Angrist, Pathak, and Walters 2013).

Previous research has documented that Boston charters boost math and English standardized test scores (Abdulkadiroğlu et al. 2011, Cohodes et al. 2013, Walters 2018). Recent evidence shows that Boston charter high schools also increase longer-term outcomes, including SAT scores, Advanced Placement (AP) credit, and enrollment in four-year college (Angrist et al. 2016). These findings are consistent with studies showing positive effects for urban No Excuses charters elsewhere (Dobbie and Fryer 2011, 2013; Angrist et al. 2010, 2012; Chabrier, Cohodes, and Oreopoulos 2016; Abdulkadiroğlu et al. 2017; Davis and Heller 2019; Winters 2020).

Funding for Massachusetts public school students follows their school enrollment. Specifically, charter schools receive tuition payments from their students' home districts equal to district per-pupil expenditure. The state partially reimburses districts for charter school payments during a transition period, but these reimbursements have not been fully funded in recent years. Prior to 2010, Massachusetts law capped the overall number of charter schools at 120 and limited total charter school tuition to 9 percent of a district's spending. Charter expenditure in Boston reached this cap in fall 2009 (Boston Municipal Research Bureau 2008). As a result, the charter cap limited the expansion of charter schools in Boston before 2010.

### **B.** Charter Expansion

In January 2010, Governor Deval Patrick signed An Act Relative to the Achievement Gap into law.<sup>6</sup> This reform relaxed Massachusetts' charter cap to allow the charter sector to double for districts in the lowest decile of performance according to a measure derived from test score levels and growth. The law also included provisions for school turnarounds and the creation of "innovation" schools (Massachusetts State Legislature 2010).

For Boston and other affected districts, the 2010 reform increased the limit on charter spending from 9 percent to 18 percent of district funds between 2010 and 2017. "Proven providers"—existing schools or school models the Massachusetts Board of Elementary and Secondary Education deemed effective—could apply to open new schools or expand enrollment. The law also allowed school districts to create up to 14 "in-district" charter schools without prior approval from the local teachers' union or proven provider status. Concurrent with the increased supply of charter seats, the law required charters to increase recruitment and retention

college preparatory curriculum, core values in school culture, selective teacher hiring or incentive pay, emphasis on math and reading, uniforms, hires Teach For America teachers, Teaching Fellows, or AmeriCorps members, affiliated with Teach For America alumni, data-driven instruction, and regular teacher feedback.

<sup>&</sup>lt;sup>6</sup> See https://malegislature.gov/Laws/SessionLaws/Acts/2010/Chapter12.

efforts for high-need students and allowed charters to send advertising mailers to all students in the district.<sup>7</sup>

The state received 71 initial applications (some of which it solicited) for new charter schools or expansions from August 2010 to August 2012, and invited 60 percent of applicants to submit final round proposals. To determine whether a school model qualified for proven provider status, the Massachusetts Board of Elementary and Secondary Education compared existing schools using the model to other charters and traditional public schools. Criteria for this evaluation included enrollment of high-need students, attrition, grade retention, dropout, graduation, attendance, suspensions, and performance on state achievement tests (Massachusetts Department of Elementary and Secondary Education 2015). The state granted proven provider status to four of seven Boston charter middle schools, as well as to the KIPP organization, which operated a charter school in Lynn, Massachusetts, but had not yet entered Boston. Together, the provisions of the 2010 reform led to the establishment of 27 new charter campuses between 2011 and 2013, as well as expansions of 17 existing charter schools, typically to new grade levels (Massachusetts Department of Elementary and Secondary Education 2016).

Charter enrollment in Boston expanded rapidly after 2010. This can be seen in Figure 1, which plots shares of kindergarten, sixth-grade, and ninth-grade students attending charter schools. These statistics are calculated using the administrative enrollment data described below. Sixth-grade charter enrollment doubled after the reform, expanding from 15 to 31 percent between 2010 and 2015. Charter enrollment also grew substantially in elementary and high school, though not as dramatically as in middle school. The share of Boston students in charter schools increased from 5 percent to 13 percent in kindergarten and from 9 percent to 15 percent in ninth grade over the same time period.

The characteristics and practices of Boston's new expansion charter schools are broadly similar to those of their proven provider parent schools. This is shown in columns 2 through 4 of Table 1, which describe proven providers, other charters operating before 2010, and new expansions. Like proven providers, expansion schools have longer school days and years than BPS schools, and rate highly on the index of No Excuses practices. Per-pupil expenditure is similar at proven provider and expansion schools, and lower at other charters. New campuses located an average of 3.1 miles from their parent campuses, often expanding into different Boston neighborhoods (see Figure 2).

Expansion charter schools are primarily staffed by young teachers with little teaching experience. Table 2 reports that 78 percent of teachers at proven providers in the year before expansion were less than 32 years old, while 87 percent of expansion charter teachers were below this threshold in the year after expansion. These and other teacher characteristics come from an administrative database of Massachusetts public school employees (see the online Data Appendix). Columns 4 and 7 show that proven providers transferred some teachers from parent campuses to help staff

<sup>&</sup>lt;sup>7</sup>The state's definition of high-need students includes those with special education or English language learner status, eligibility for subsidized lunch, or low scores on state achievement tests, as well as students deemed to be at risk of dropping out of school.



FIGURE 1. CHARTER SCHOOL ENROLLMENT IN BOSTON

*Notes:* This figure plots the share of Boston fourth-, sixth-, and ninth-grade students enrolled in charter schools between the 2001–2002 and 2014–2015 school years. The gray dashed line denotes the last school year before the charter expansion policy went into effect.



FIGURE 2. LOCATIONS OF BOSTON CHARTER SCHOOLS

*Notes:* This figure maps the location of the middle school charters in Boston, including schools that expanded (proven providers), new charter schools (expansion charters), and other charters. Each color denotes a different charter network.

their expansions: 12 percent of parent campus teachers moved to expansion campuses, accounting for 25 percent of the teaching workforce at these new schools.

		Tea	Teachers at proven providers in 2010–2011				Teachers at expansion charters in first year		
	BPS overall (1)	All (2)	Stay at parent campus (3)	Move to expansion (4)	Leave network (5)	All (6)	Came from parent campus (7)	New teacher (8)	
Fraction in category		1.00	0.62	0.12	0.26	1.00	0.25	0.66	
<32 years old	0.30	0.78	0.73	0.95	0.85	0.87	0.86	0.89	
>49 years old	0.26	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Unlicensed	0.04	0.28	0.24	0.29	0.38	0.53	0.07	0.76	
Years working in Massachusetts public schools	11.47	2.89	3.26	2.20	2.25	1.44	3.41	0.45	
Observations (full time equivalent teachers)	4,261	88	54	11	22	55	14	36	

TABLE 2—STAFFING AT PROVEN PROVIDER AND EXPANSION CHARTER SCHOOLS

*Notes:* This table describes characteristics of teachers at Boston charter schools before and after expansion. Column 1 summarizes Boston Public Schools (BPS) teacher characteristics in 2011–2012. Columns 2–5 display statistics for teachers working at proven provider charters in the 2010–2011 school year. Columns 6–8 show statistics for teachers working at expansion charters during the 2011–2012 school year. New teacher status in column 8 is defined as having less than one year of experience teaching in Massachusetts public schools. A small number of expansion charter teachers came from schools other than the parent campus and their characteristics are similar to teachers in column 7.

Transferred teachers were less experienced than teachers who remained at parent campuses (2.2 years versus 3.3 years). Most of the remaining expansion teachers had not previously taught in a Massachusetts school (66 percent), though a few transferred from other schools (9 percent). As a result, the average teacher at an expansion charter had only 1.4 years of teaching experience, compared to 2.9 years for teachers at parent campuses and 11.5 years for BPS teachers.

### II. Data

#### A. Data Sources and Sample Construction

We study the effectiveness of Boston charter middle schools using records from randomized admission lotteries conducted between 2004 and 2013. Some charters serving middle school grades (fifth through eighth) accept students prior to fifth grade, mostly in kindergarten; we focus on schools with fifth- or sixth-grade entry because their lotteried students are old enough to take achievement tests within our data window. Our sample includes 14 of the 15 Boston charter schools with fifth- or sixth-grade entry, accounting for 94 percent of enrollment for schools in this category during the 2013–2014 school year.<sup>8</sup>

Lottery records typically list applicant names along with application grades, dates of birth, towns of residence, and sibling statuses. Our analysis excludes sibling applicants, out-of-area applicants, and students who applied to nonentry grades (siblings are guaranteed admission, while out-of-area applicants are typically ineligible). The lottery records also indicate which students received admission

<sup>&</sup>lt;sup>8</sup>Two charter middle schools that closed before 2010 are excluded from this calculation. The one missing school declined to provide lottery records.

offers. We distinguish between immediate offers received on the day of the lottery and later offers received from the waitlist; in some lotteries, all students eventually receive waitlist offers. All offers are coded as waitlist offers in a few lotteries where we cannot distinguish between immediate and waitlist offers. Further information on school coverage and lottery records appears in online Appendix Tables A1 and A2. We use the "proven provider" label to refer to the four middle school charters in Boston that were granted permission to expand. The seven new campuses opened in the 2011–2012 and 2012–2013 school years are labeled "expansion charters," and the three remaining charter middle schools are "other charters."<sup>9</sup>

We match the lottery records to state administrative data based on name, date of birth, town of residence, and application cohort. The administrative data cover all students enrolled in Massachusetts public schools between 2002 and 2014. As shown in online Appendix Table A3, we find matches for 95 percent of lottery applicants in this database. Administrative records include school enrollment, gender, race, special education status, English language learner status, subsidized lunch status, and test scores on Massachusetts Comprehensive Assessment System (MCAS) achievement tests. We standardize MCAS scores to have mean zero and standard deviation one for Boston students by subject, grade, and year. In addition to information on charter lottery applicants, we use administrative data on other Boston students to describe changes in charter application and enrollment patterns after the 2010 reform. The online Data Appendix provides more details regarding data processing and sample construction.

### **B.** Descriptive Statistics

Charter application and enrollment patterns in our analysis sample mirror the large increases in charter market share displayed in Figure 1. As shown in Table 3, 15 percent of eligible Boston students applied to charter schools with fifth- or sixthgrade entry before the 2010 reform, 12 percent received offers from these schools, and 10 percent enrolled. This implies roughly 1.5 applicants for each available charter seat. The application rate increased to 35 percent in 2013, and attendance reached 17 percent.<sup>10</sup> The increase in applications therefore outpaced enrollment growth, boosting the number of applicants per seat to 2. This increase in demand was particularly pronounced at other charter schools (neither proven providers nor expansions), which saw their applications per seat rise from 1.9 to 4.<sup>11</sup> After

<sup>&</sup>lt;sup>9</sup>We categorize MATCH Middle School as a proven provider, as MATCH obtained that categorization from the state. MATCH's expansion campus opened at the elementary level, however. We categorize KIPP: Boston as an expansion campus, but this school does not have a direct parent campus in Boston as KIPP's only previous Massachusetts campus was in Lynn. We classify UP Academy as an expansion charter even though it opened under a different provision of the charter school law. To check whether our results are sensitive to these classification decisions, online Appendix Table A4 reports an alternative version of our main results with these three schools categorized as "other charters." The findings here remain generally the same, with other charters demonstrating slightly larger effects and proven providers and expansion schools showing smaller gains than in our preferred specification.

<sup>&</sup>lt;sup>10</sup>These attendance percentages are lower than the percentages in Figure 1, since they exclude charter schools that enroll students at earlier entry grade levels.

<sup>&</sup>lt;sup>11</sup>The number of applicants per seat is larger for each individual charter type than for the sector as a whole because some students apply to more than one school.

	Befor	e charter exp	oansion	After charter expansion			
	Any charter (1)	Proven providers (2)	Other charters (3)	Any charter (4)	Proven providers (5)	Expansion charters (6)	Other charters (7)
Percent of Boston	15	9	8	35	19	19	18
Percent of Boston students with lottery offers	4	2	3	10	4	7	3
Percent of Boston students with lottery or waitlist offers	12	7	6	23	10	15	6
Percent of Boston students enrolling in charters	10	5	4	17	5	9	4
Applicants per seat	1.5	1.8	1.9	2.0	3.4	2.2	4.0

TABLE 3—CHARTER MIDDLE SCHOOL APPLICATIONS AND ENROLLMENT

*Notes:* This table summarizes applications and enrollment for Boston charter middle schools in the analysis sample before and after the 2010–2011 charter sector expansion. The sample of charters excludes schools serving middle school grades with primary entry points prior to fifth grade. Students are included if they enrolled in Boston schools in both fourth and sixth grade. Columns 1–3 show statistics for cohorts of students entering fifth grade in fall 2008 or 2009. Columns 4–7 display statistics for cohorts entering fifth grade in fall 2011–2013.

TABLE 4—CHARACTERISTICS OF BOSTON MIDDLE SCHOOL STUDENTS

	Before charter expansion			After charter expansion				
	Enrolled	Rande appl	omized icants	Enrolled	Ran	Randomized applicants		
	BPS (1)	All charters (2)	Proven providers (3)	BPS (4)	All charters (5)	Proven providers (6)	Expansion charters (7)	
Female	0.478	0.487	0.485	0.476	0.493	0.484	0.486	
Black	0.418	0.561	0.638	0.313	0.443	0.450	0.453	
Latino/a	0.353	0.237	0.295	0.435	0.406	0.453	0.432	
Asian	0.093	0.018	0.012	0.096	0.033	0.025	0.034	
White	0.122	0.171	0.040	0.130	0.092	0.047	0.053	
Subsidized lunch	0.839	0.687	0.742	0.792	0.802	0.835	0.831	
English language learners	0.223	0.117	0.160	0.410	0.363	0.412	0.395	
Special education	0.248	0.191	0.181	0.236	0.202	0.197	0.209	
Attended charter in 4th grade	0.002	0.120	0.093	0.001	0.040	0.028	0.016	
4th grade math score		0.220	0.043		0.021	0.020	-0.061	
4th grade English score	—	0.303	0.156	—	0.023	-0.014	-0.090	
Observations	18,934	2,724	1,263	8,330	4,478	2,250	2,414	

*Notes:* This table shows descriptive statistics for Boston middle school students before and after the 2010–2011 charter school sector expansion. The sample includes all students who attended Boston schools in fourth grade and fifth or sixth grade between 2004 and 2013. Columns 1 and 4 show statistics for students who did not enroll in a charter school in fifth or sixth grade. Columns 2, 3, 5, 6, and 7 report statistics for randomized charter school applicants. Randomized applicants exclude siblings, disqualified students, and out-of-area applicants. Test scores are standardized to have mean zero and standard deviation one in Boston by subject, grade, and year.

the expansion, half of charter school sixth-grade students attended new expansion campuses.

Table 4 describes the characteristics of Boston middle school students in BPS and our randomized lottery applicant sample. Charter applicants are consistently more likely to be Black than BPS students. Both before and after 2010, students attending proven providers were less disadvantaged than other Boston students as measured by special education status, English learner status, and fourth-grade test scores. As shown in Table 4, the characteristics of applicants to expansion charters differ markedly from those of other charter students. Special education and English learner rates are similar among expansion charter applicants and the BPS population. Expansion charter applicants also score below the BPS average on fourth-grade math and English tests, and are more likely than BPS students to be eligible for subsidized lunches. These facts indicate that expansion charters attract a more disadvantaged, lower-achieving population than their proven provider parent schools. This pattern may reflect the changes in recruitment practices resulting from the 2010 Achievement Gap Act, which mandated that charter schools take steps to enroll higher-need students and allowed charters to advertise directly to all students in the district by mail.

#### **III. Empirical Framework**

We use charter lottery offers as instruments for charter school attendance in a causal model with multiple endogenous variables, each representing enrollment in a type of charter school. The structural equation links charter attendance with outcomes as follows:

(1) 
$$Y_{ig} = \alpha_g + \sum_{k=1}^K \beta_k C_{ig}^k + \sum_{j=1}^J \delta_j R_{ij} + X_i' \gamma + \varepsilon_{ig},$$

where  $Y_{ig}$  is a test score for charter applicant *i* in grade *g* and  $C_{ig}^k$  measures years of enrollment in charter school type *k* through grade *g*.<sup>12</sup> Charter types include parent campuses, expansion campuses, and other charters; we also distinguish between enrollment before and after the charter expansion law. The parameters of interest,  $\beta_k$ , represent causal effects of an additional year of attendance at each charter type relative to traditional public schools.<sup>13</sup> The key control variables in equation (1) are a set of indicators,  $R_{ij}$ , for all combinations of charter lottery applications present in the data. Lottery offers are randomly assigned within these "risk sets." A vector of baseline demographic characteristics,  $X_i$ , is also included to increase precision. These characteristics, which are measured in the year prior to a student's lottery application, include gender, race, a female-minority interaction, subsidized lunch status, English language learner status, and special education status.

The first-stage equations for each charter enrollment type are given by

(2) 
$$C_{ig}^{k} = \mu_{g}^{k} + \sum_{\ell=1}^{K} \left( \pi_{\ell 1}^{k} Z_{i1}^{\ell} + \pi_{\ell 2}^{k} Z_{i2}^{\ell} \right) + \sum_{j=1}^{J} \lambda_{j}^{k} R_{ij} + X_{i}^{\prime} \theta^{k} + \eta_{ig}^{k}; \quad k = 1, \dots, K.$$

Here,  $Z_{i1}^k$  denotes a dummy variable equal to one if applicant *i* received an immediate offer to attend charter type *k* on the day of a lottery, and  $Z_{i2}^k$  equals one if the applicant later received an offer from the waitlist. Like equation (1), the first stage also controls for lottery risk set indicators and baseline student characteristics.

<sup>&</sup>lt;sup>12</sup>Test scores are the first instance that a student takes the MCAS in that grade. Years of enrollment includes repeated grades.

<sup>&</sup>lt;sup>13</sup> If treatment effects vary across students or years of attendance these coefficients can be interpreted as average causal responses (ACRs), weighted averages of causal effects for individuals whose attendance is shifted by lottery offers (Angrist and Imbens 1995).

Two-stage least squares (2SLS) estimates are obtained by ordinary least squares (OLS) estimation of equation (1) after substituting predicted values from (2) for the charter attendance variables. The estimation sample stacks all post-lottery test scores in grades five through eight for randomized charter applicants, and standard errors are clustered by student to account for correlation in outcomes across grades.

Our empirical strategy is motivated by the fact that charter lottery offers are randomly assigned within lottery risk sets and are therefore independent of ability, family background, and all other predetermined student attributes. Online Appendix Table A5 presents a check on this by comparing baseline characteristics for offered and nonoffered applicants within lottery risk sets. These comparisons show that lottery winners and losers are similar for all charter school types and time periods, indicating that random assignment was successful.<sup>14</sup>

### **IV. Lottery Estimates**

Proven provider charter schools generated larger achievement gains than other charter schools in Boston prior to the 2010 expansion. This can be seen in Table 5, which reports two-stage least squares estimates of equations (1) and (2).<sup>15</sup> The first-stage estimates in panel A show that charter offers boosted years enrolled by about one year before expansion, and around half a year after expansion. This reflects the fact that less time has elapsed in our data for cohorts applying after 2010, resulting in fewer years of potential charter enrollment between lottery and test dates. Columns 2 and 3 of panel B demonstrate that a year of charter attendance at a proven provider increased math and English scores by  $0.32\sigma$  and  $0.12\sigma$  prior to the reform, estimates that are highly statistically significant. Corresponding math and English effects for proven providers and other charters is statistically significant in math (p = 0.00), though not in English. This finding indicates that policymakers selected more effective charter schools to be eligible for expansion.

Proven providers and other charters maintained their effectiveness after the charter expansion reform. As shown in columns 5 and 7 of Table 5, proven providers boost math and English scores by  $0.37\sigma$  and  $0.19\sigma$  per year of attendance after 2010, while other charters increase scores by  $0.19\sigma$  and  $0.13\sigma$  in this period. These estimates are slightly larger than estimates for earlier cohorts, though the differences between pre- and post-reform effects are not statistically significant for either group. As in the pre-reform period, the difference in effects between proven providers and other charters is significant in math (p = 0.03). These results indicate that

<sup>&</sup>lt;sup>14</sup>Even with random assignment, selective attrition may lead to bias in comparisons of lottery winners and losers. Online Appendix Tables A3 and A6 show that the attrition rate from our sample is low: we match 95 percent of applicants to the administrative data, and find roughly 85 percent of post-lottery test scores that should be observed in our sample window for matched students. The match rate is 4 percent higher for students offered charter seats, and we are 3 percent more likely to find scores for students with lottery offers at non-proven-provider charters before 2010. This modest differential attrition seems unlikely to meaningfully affect the results reported below.

<sup>&</sup>lt;sup>15</sup> Online Appendix Table A7 reports a pooled set of 2SLS estimates combining charter types and time periods. Reduced-form estimates are reported in online Appendix Table A8 and OLS estimates of charter school effects that control for prior test scores and baseline characteristics appear in online Appendix Table A9.

	Before charter expansion			After charter expansion				
		Estin	nates		Estimates			
	Non-charter mean (1)	Proven providers (2)	Other charters (3)	Non-charter mean (4)	Proven providers (5)	Expansion charters (6)	Other charters (7)	
Panel A. First-stage estima	tes							
Immediate offer		1.304 (0.067)	1.554 (0.047)		$0.795 \\ (0.054)$	0.659 (0.046)	$0.930 \\ (0.052)$	
Waitlist offer		$1.027 \\ (0.050)$	$\begin{array}{c} 0.984 \\ (0.061) \end{array}$		$\begin{array}{c} 0.400 \\ (0.048) \end{array}$	$\begin{array}{c} 0.348 \ (0.041) \end{array}$	$\begin{array}{c} 0.853 \\ (0.071) \end{array}$	
Panel B. 2SLS estimates								
Math	0.117	0.320 (0.037)	0.183 (0.026)	-0.074	$0.365 \\ (0.070)$	$0.326 \\ (0.074)$	$0.193 \\ (0.055)$	
<i>p</i> -value: Equals proven provider			0.000			0.632	0.030	
<i>p</i> -value: Equals other charters						0.070		
Observations (applicants) Observations (total scores)	1,093	1,279	1,909	2,443 17,395	2,303	2,416	2,405	
English	0.201	0.122 (0.037)	0.084 (0.025)	-0.032	$0.186 \\ (0.074)$	$0.229 \\ (0.076)$	$\begin{array}{c} 0.126 \\ (0.054) \end{array}$	
<i>p</i> -value: Equals proven provider			0.324			0.619	0.470	
<i>p</i> -value: Equals other charters						0.162		
Observations (applicants) Observations (total scores)	1,087	1,277	1,911	2,441 17,316	2,307	2,420	2,412	

TABLE 5—CHARTER EFFECTS ON TEST SCOP	RES BEFORE AND AFTER CHARTER EXPANSION
--------------------------------------	--

*Notes:* Panel A reports first-stage effects of charter lottery offers on years of enrollment in charter schools. Panel B displays 2SLS estimates of the effects of charter school attendance on test scores. The sample stacks post-lottery test scores in grades five through eight. The endogenous variables are counts of years spent in the different charter types (pre-expansion proven providers, pre-expansion other charters, post-expansion proven providers, expansion schools, and post-expansion other charters). The instruments are immediate and waitlist lottery offer dummies for each school type. Immediate offer equals one for applicants offered seats on the day of the lottery. Waitlist offer equals one for applicants offered seats from the waitlist. Controls include lottery risk sets, as well as gender, race, ethnicity, a female-minority interaction, special education, English language learner, subsidized lunch status, and grade and year indicators. Standard errors are clustered by student.

expanding to operate new campuses did not dilute the effectiveness of proven provider charters at their original campuses.

Proven providers also successfully replicated their impacts at expansion schools. Column 6 of Table 5 demonstrates that a year of attendance at an expansion charter school increases math and English test scores by  $0.32\sigma$  and  $0.23\sigma$ . These estimates are comparable to estimates for parent campuses, and the hypothesis that expansion and proven provider effects are equal cannot be rejected at conventional levels (p = 0.63 and 0.62 in math and English). Estimated effects for expansion charters are larger than corresponding estimates for other charters during the same time period, though these differences are only marginally statistically significant for math and not statistically significant for English. Combined with the consistent effects for proven providers and other charters over time, these results indicate that Boston's charter middle school sector slightly increased its average effectiveness despite the growth in charter market share over this period. These findings are robust to a number of specification checks. The results are generally similar when test scores are limited to the first year after entrance to a charter, though standard errors increase due to the reduction in power (online Appendix Table A10). Focusing on the first year addresses the concern that charter students might spend more time in grade before their exams due to grade retention.<sup>16</sup> Additionally, if charter impacts varied a great deal by grade level, the cross-period findings could be influenced by a different mix of grade levels, since there is a longer time horizon in the pre-expansion period. By limiting the findings to the year after the lottery, both the pre- and post-expansion groups outcomes are limited to fifth- and sixth-grade scores. In the truncated sample in math, proven providers outperform expansion campuses, though expansion campuses still have very large impacts and the difference is marginally significant. In English, the opposite is true. Both proven providers and expansion campuses retain their edge over other charter campuses.

Another specification check addresses issues raised by de Chaisemartin and Behaghel (2020), who argue that in cases with small lottery samples, the "waitlist offer" instrument can generate bias because the student receiving the final "waitlist offer" is different than the average waitlist member (as they respond affirmatively to the offer). Waitlists in our context are generally large, but we still address this concern by showing estimates that only use the initial offer on the day of the lottery as an instrument in online Appendix Table A11. As predicted, there are few differences in our findings, though we need to exclude a few lotteries where only waitlist offer information was retained. If anything, the charter effects are slightly larger in a specification with the initial offer-only instrument.

Finally, we also consider whether changes in peer quality or school switching are driving the charter findings. Panel A of online Appendix Table A12 shows 2SLS estimates of impacts on the baseline test scores of school-level peers; panel B shows 2SLS estimates of the likelihood of switching schools. Charter attendance boosts the likelihood that students attend school with higher-achieving peers. However, peer gains by charter type do not align with achievement gains. Other charters produce the largest gains in peer quality but the smallest impacts on test scores. Expansion campuses produce small changes in peer quality but large test score effects. The charter school peer advantage also diminishes over time; by the third year after the lottery, we find only small effects on peer quality in the pre-expansion period and minimal effects in the post-expansion period. Angrist et al. (2016) document a similar pattern for charter high schools. Furthermore, even in the first year, the peer effect would have to be *larger* than the peer differential to account for the full magnitude of the charter effects. The peer effects literature typically finds that peer effects transmit at a 10 to 30 percent rate (Sacerdote 2011). These results suggest that changes in peer quality are not the channel mediating charter gains.

Charter schools' effectiveness may be driven in part by decisions to push disruptive students out of schools. Our impact estimates account for this by assigning a student a full year of charter attendance even if he or she only attends a charter

<sup>&</sup>lt;sup>16</sup>We also restrict test scores to the first in-grade exam and count years of charter attendance to include repeaters in all specifications.

for one day during the year. However, a direct investigation of effects on school switching is also of interest in view of concerns about selective push-out at charter schools. Panel B of online Appendix Table A12 reveals that winning a charter lottery reduces the likelihood that a student subsequently switches schools. This effect is partially due to differences in transition grades, as BPS students typically transition from elementary school to middle school in sixth grade whereas some charter middle schools start in fifth grade. When we examine school switching at nontransition grades, however, charter students remain 6 to 9 percentage points less likely to switch schools (though these differences are generally not statistically significant in the post-expansion period).

### V. Exploring Effect Heterogeneity

Massachusetts' charter expansion reform led to a larger increase in Boston's charter market share than expansions evaluated in previous studies, which typically look at diffuse growth of charter networks across many markets (e.g., Tuttle et al. 2015). This suggests that mechanisms related to Boston's uniquely large within-market expansion may be important for understanding the effects of the reform. Charter skeptics commonly argue that charters succeed by "cream skimming" small numbers of unusually motivated students (Rothstein 2004). A large within-market expansion necessarily requires charters to enroll a new population of students, which may limit the scope for such cream skimming and change the mix of students that selects into the charter sector more generally. The role of student selection is especially policy relevant here since Massachusetts' expansion law encouraged charters to recruit and retain students with higher needs, as measured by criteria including English proficiency, special education status, and past achievement. Relatedly, a large literature also argues that school choice programs may affect the performance of traditional public schools, either through cream skimming and negative peer effects (e.g., Ladd 2002) or through competition that pressures public schools to improve (Hoxby 2003). We next investigate these potential mechanisms by exploring effect heterogeneity across students and fallback traditional public schools.

### A. Student Characteristics

As a starting point for our investigation of student selection, online Appendix Table A13 summarizes effect heterogeneity as a function of observed student characteristics. The estimates show consistent positive impacts across most subgroups, charter school types, time periods, and subjects. Effects are similar for English language learners and students without this designation, though estimates for the former group are often imprecise due to small sample sizes. All estimates are positive for students with and without special education status; effects for special education students appear to be somewhat smaller at proven providers and larger at expansion charters, but these differences may be a chance finding due to the many splits examined. As in previous studies (e.g., Walters 2018), we find that effects tend to be larger for students with lower previous test scores. The large estimated effects for high-need subgroups at expansion charters are noteworthy: expansion schools

continue to generate substantial gains for these groups despite serving larger shares of such students than other Boston charters.

We analyze the consequences of this heterogeneity for the effectiveness of charter expansion via a Oaxaca-Blinder-style decomposition, which splits charter school treatment effects into components explained and unexplained by student characteristics (Oaxaca 1973, Blinder 1973). This decomposition is based on 2SLS estimates of the equation

(3) 
$$Y_{ig} = \alpha_g + \sum_{k=1}^{K} \left(\beta_k^0 + X'_i \beta^x\right) C_{ig}^k + \sum_{j=1}^{J} \delta_j R_{ij} + X'_i \gamma + \varepsilon_{ig}.$$

Equation (3) allows a separate main effect for attendance at each charter type  $(\beta_k^0)$  as well as an interaction with student characteristics common across charter types  $(\beta^x)$ . Charter exposure  $C_{ig}^k$  and its interactions with  $X_i$  are treated as endogenous. The immediate and waitlist offer variables for each charter type  $Z_{i1}^k$ ,  $Z_{i2}^k$ , and their interactions with  $X_i$  are the excluded instruments.

Let  $\bar{X}_k$  denote the average characteristics of students attending charter k, and let  $\mu^x \equiv E[X_i]$  denote the mean of  $X_i$  for the Boston population. Assuming that all treatment effect heterogeneity is captured by observed characteristics, the effect of charter type k for students enrolled at k (the effect of treatment on the treated,  $TOT_k$ ) can be represented as

(4)  
$$TOT_{k} = \beta_{k}^{0} + \bar{X}_{k}^{\prime}\beta^{x}$$
$$= \underbrace{\left(\beta_{k}^{0} + \mu^{x\prime}\beta^{x}\right)}_{ATE_{k}} + \underbrace{\left(\bar{X}_{k} - \mu^{x}\right)^{\prime}\beta^{x}}_{Match_{k}}.$$

This expression decomposes the TOT for charter type k into an average treatment effect for the Boston population,  $ATE_k$ , and a deviation from the average treatment effect due to the characteristics of type k's students,  $Match_k$ . If  $Match_k > 0$ , students with atypically high benefits select into the charter sector, while  $Match_k < 0$  would imply that charter students benefit less than the average Boston student. We might expect a large charter expansion to reduce  $Match_k$  by drawing in new students who, at the margin, benefit less from charter attendance than more eager students who attended when the sector was small. On the other hand, Walters (2018) argues that in earlier periods Boston's charter sector attracted students with lower than average gains, perhaps because the intensive charter treatment is more helpful for those with less motivated parents who are also less likely to seek alternative schooling options. We assess these ideas by studying estimates of  $ATE_k$  and  $Match_k$  for each school type and time period.

Table 6 reports estimates of the components of the decomposition in equation (4) using gender, race, ethnicity, English language learner status, subsidized lunch, special education, and baseline test scores as interaction variables. Two-stage least squares estimates appear in panel A, and panel B displays results based on OLS estimates of equation (3) for comparison. As with the treatment effect estimates in online Appendix Table A9, the OLS decomposition results tend to be qualitatively similar and more precise than the 2SLS results. Estimated match components are

	Before expan	charter nsion	Afte	After charter expansion			
	Proven providers (1)	Other charters (2)	Proven providers (3)	Expansion charters (4)	Other charters (5)		
Panel A. 2SLS results							
Math							
TOT	0.333	0.185	0.319	0.359	0.197		
	(0.029)	(0.020)	(0.050)	(0.052)	(0.037)		
ATE	0.320	0.198	0.321	0.345	0.208		
	(0.030)	(0.022)	(0.051)	(0.053)	(0.038)		
Match	0.013	-0.013	-0.002	0.014	-0.011		
	(0.009)	(0.009)	(0.008)	(0.005)	(0.005)		
Observation (scores)			15,924				
English							
TOT	0.185	0.100	0.156	0.207	0.096		
	(0.030)	(0.020)	(0.053)	(0.051)	(0.039)		
ATE	0.180	0.119	0.144	0.190	0.105		
	(0.031)	(0.022)	(0.054)	(0.052)	(0.040)		
Match	0.004	-0.019	0.013	0.016	-0.009		
	(0.009)	(0.009)	(0.008)	(0.005)	(0.005)		
Observations (scores)			15,932				
Panel B. OLS results Math							
TOT	0.365	0.234	0.307	0.326	0.228		
	(0.009)	(0.009)	(0.011)	(0.014)	(0.011)		
ATE	0.361	0.258	0.306	0.313	0.243		
	(0.010)	(0.010)	(0.011)	(0.014)	(0.011)		
Match	0.003	-0.023	0.001	0.013	-0.015		
	(0.003)	(0.003)	(0.003)	(0.002)	(0.002)		
Observations (scores)		:	84,246				
English							
ТОТ	0.275	0.094	0.203	0.164	0.200		
	(0.010)	(0.010)	(0.011)	(0.013)	(0.010)		
ATE	0.280	0.125	0.191	0.149	0.215		
	(0.010)	(0.011)	(0.012)	(0.013)	(0.010)		
Match	-0.004	-0.031	0.012	0.015	-0.015		
	(0.003)	(0.003)	(0.003)	(0.002)	(0.002)		
Observations (scores)		:	84,290				

TABLE 6—DECOMPOSITION OF CH	ARTER SCHOOL EFFECTS
-----------------------------	----------------------

*Notes:* This table decomposes estimates of charter school treatment effects into components explained and unexplained by student characteristics. These characteristics are female, Black, Latino/a, subsidized lunch, English language learner, special education, and baseline test scores. Estimates in panel A come from 2SLS models treating years of enrollment in each charter type and years in any charter interacted with student characteristics as endogenous, instrumenting with charter lottery offers and their interactions with student characteristics. These models control for main effects of student characteristics and lottery risk sets, and are estimated in the sample of randomized applicants. Estimates in panel B come from corresponding OLS models estimated in the full sample of Boston students. These models exclude lottery risk sets and include controls for Asian, non-white other race, baseline charter attendance, and a female-minority interaction.

close to zero for proven providers in both time periods, while match components for other charters are negative in both periods. This indicates that the demographic composition of other charters reduces their effectiveness, a result that is consistent with Walters' (2018) finding that disadvantaged students were less likely to apply to Boston charter schools despite experiencing larger achievement benefits in data prior to the reform.

In contrast, column 4 of Table 6 reveals positive match effects for expansion charter schools. This pattern is due to the fact that expansion charters enroll a lower-achieving set of students compared to other charters (see Table 4). Since achievement gains are larger for this group, the match effect reinforces the effectiveness of expansion charters. The magnitudes of these match effects are relatively small, however, accounting for roughly 4 percent and 8 percent of the TOT in math and ELA. Changes in student characteristics increased the effectiveness of expansion schools.

### **B.** Fallback Schools

One potential explanation for the success of Boston charter school expansion, where other efforts at program replication have been less successful, is that students in expansion campuses face particularly poor alternatives if they do not attend a charter school. Chabrier, Cohodes, and Oreopolous (2016) find that poor fallback school options are one of the strongest predictors of charter school effectiveness. It is also possible that charter schools influence the counterfactual by diverting resources from district schools (Arsen and Ni 2011, Bifulco and Reback 2014, Cook 2018, Ladd and Singleton 2020). However, Ridley and Terrier (2018) find small gains in district school finances (and test scores) in Massachusetts using the same charter expansion law.<sup>17</sup> Charter operators may have intentionally opened expansion campuses in areas of Boston with lower-performing traditional public schools. To see if low-quality fallback schools explain the success of expansion campuses, we compare fallback school conditions across charter school types, both before and after charter school expansion.

Table 7 shows average school-level value-added estimates for traditional public schools attended by students that enroll in district schools as a result of losing a charter lottery (untreated compliers).<sup>18</sup> Value-added estimates are OLS coefficients from regressions of test scores on school indicators, with controls for lagged test scores and demographics. Specification tests reported by Angrist et al. (2017) indicate that estimates from models of this type provide a reasonable proxy for school effectiveness. In both math and English, estimated value-added of the traditional public school fallback alternatives attended by charter applicants does not differ by charter school type, and these fallback schools appear to be of roughly average quality among schools in BPS. Students' fallback options

<sup>&</sup>lt;sup>17</sup>Other studies of competitive effects of charter schools on nearby district schools' test scores generally find zero or small positive impacts (Booker et al. 2008; Cordes 2018; Jinnai 2014; Davis 2013; Sass 2006; Shin, Fuller, and Dauter 2017; Winters 2012; Zimmer et al. 2009; Zimmer and Buddin 2009). One exception is Imberman (2011), which found a mix of neutral and negative effects. For reviews of this literature, see Betts (2009); Gill and Booker (2008); Gill (2016); and Epple, Romano, and Zimmer (2016).

<sup>&</sup>lt;sup>18</sup> We estimate untreated complier outcomes using methods from Abadie (2002).

	Before expan	charter 1sion	Afte	r charter expan	ision
	Proven providers (1)	Other charters (2)	Proven providers (3)	Expansion charters (4)	Other charters (5)
Untreated complier mean: math	0.008 (0.010)	0.015 (0.009)	0.028 (0.015)	0.017 (0.017)	0.027 (0.013)
Observations		7,	194		
Untreated complier mean: English	-0.015 (0.008)	-0.012 (0.008)	$0.000 \\ (0.013)$	-0.007 (0.015)	-0.007 (0.012)
Observations		7,	194		

TABLE 7—VALUE-ADDED OF FALLBACK SCH	100LS FOR CHARTER APPLICANTS
-------------------------------------	------------------------------

*Notes:* This table summarizes OLS value-added estimates for schools attended by untreated charter lottery compliers. Untreated complier means are estimates from 2SLS regressions of school value-added interacted with a traditional public school indicator on a set of variables equal to one minus attendance at each charter type, instrumented with charter lottery offers and controlling for demographics and lottery risk sets. School value-added estimates come from OLS regressions of test scores on a set of school indicator variables, controlling for lagged test scores and student demographics.

therefore do not seem to be an important component of variation in effects across charter types or time periods.

### VI. School Practices

Our results so far show that changes in student characteristics and the quality of applicants' fallback schools do not explain the effectiveness of expansion charters. This suggests that successful replication of the Boston charter model may be driven by attributes of the expansion schools themselves. We explore this hypothesis by providing a more detailed account of organizational practices at parent and expansion charter schools in Boston. This portion of our analysis includes a qualitative overview of the mechanics of charter expansion based on interviews with school leaders,<sup>19</sup> as well as a quantitative assessment of teacher value-added that gives an indication of how heterogeneity in teacher quality is managed in traditional public and charter schools.

### A. Standardized School Models and Leadership

Proven provider charter schools sought to maintain fidelity of their school models during expansion by emphasizing adherence to the same educational practices at new campuses. Table 1 shows a comparison of practices at parent and expansion charters based on information drawn from charter school annual reports.<sup>20</sup>

<sup>&</sup>lt;sup>19</sup> S. Dunn, J. Clark, W. Austin, A. Hall, and D. Lehman, personal communication, May 2017.

<sup>&</sup>lt;sup>20</sup>The Massachusetts Department of Elementary and Secondary Education provided the 2012–2013 annual reports for each of the Boston charter middle schools at our request. The state requires charter schools to submit annual reports and uses the reports when considering schools' charter renewal applications.

Expansion schools typically have the same amount of instructional time as their parent campuses, including identical length of the school day, time devoted to math and reading instruction, and days in the school year.<sup>21</sup> Expansion schools similarly implemented their parent campuses' No Excuses practices, tutoring, homework help, and Saturday school programs.

Expanding charter networks also tried to maintain similar pedagogical practices at old and new campuses. Teachers co-planned curricula, and teachers judged to be effective were encouraged to share their lesson plans across the network. This model of shared teaching resources was aimed at supporting new, inexperienced teachers, who comprised two-thirds of the new schools' staff. Survey evidence from Boston charters indicates that such collaboration is common within the sector, with 59 percent of new teachers reporting co-planning the curriculum with their peers (The New Teacher Project 2014). Recent evidence from other contexts shows that such collaboration can increase student achievement (Jackson and Bruegmann 2009; Ronfeldt et al. 2015; Papay et al. 2020; Sun, Loeb, and Grissom 2017) and that access to high-quality lesson plans also boosts student achievement (Jackson and Makarin 2018).

High teacher turnover rates are the norm at Boston charter schools. This is shown in Table 8, which summarizes teacher mobility patterns at charter and traditional public schools. As a result, some practices aimed at quickly training new teachers were in place prior to the 2010 reform. This may have aided schools' efforts to bring inexperienced teachers at new campuses up to speed on key practices. Two charter networks run their own teacher training programs and hired some of the graduates as full-time teachers. Charter networks also centralized teacher recruitment and professional development, potentially saving on search costs and resulting in similar types of teachers hired at new and old schools. Each network reported conducting some share of professional development at the network level.

Growing charter networks had stable leadership throughout the scaling-up process. Principals in the new and original campuses did not change throughout the expansion period in this study.<sup>22</sup> Furthermore, principals were trained internally: all of the principals at expansion campuses were former teachers from the original campus. School leaders who oversaw their networks' expansions stressed the value of selecting principals from within the network because of their familiarity with core school practices. Columns 3 and 6 of Table 8 show that roughly 4 percent of charter school teachers were promoted to a leadership position from 2011 to 2014, compared to less than 1 percent of BPS teachers.

### B. Evidence on Teacher Productivity

The qualitative evidence above suggests that Boston charter schools limit teacher discretion by emphasizing a standard set of pedagogical practices, which may

<sup>&</sup>lt;sup>21</sup>Edward Brooke's replication campus in East Boston is an exception, with six more days in its school year than its parent campus.

<sup>&</sup>lt;sup>22</sup> We verified this in Education Personnel Information Management Systems (EPIMS), the educator database available from the Department of Elementary and Secondary Education, which contains yearly staff-level data for all employees in Massachusetts public schools.

		BPS		Charters			
	New teachers (<5 years of experience) (1)	Experienced teachers (2)	All (3)	New teachers (<5 years of experience) (4)	Experienced teachers (5)	All (6)	
Panel A. Year-to-year teacher mo	bility						
Percent remain teachers at school	70.9	78.9	76.4	68.5	76.1	69.7	
Percent stop teaching at school	29.1	21.1	23.6	31.5	23.9	30.3	
Observations	1,732	3,916	5,629	952	176	1,127	
Panel B. Destinations for teacher	s who leave						
Percent teach at another school	13.4	10.1	11.2	7.6	9.7	7.9	
Percent leave teaching	18.4	13.2	14.8	24.5	14.2	22.8	
Percent become school leader	0.4	0.7	0.6	3.8	6.3	4.2	
Observations	558	940	1,495	341	53	393	

TABLE 8—TEACHER MOVEMENT

*Note:* This table summarizes the year-to-year changes in employment of teachers who taught in Boston Public Schools (BPS) or charter schools between 2010 and 2013. Panel A displays work status in the following year for each teacher-year observation. Panel B displays destinations for teachers who switched schools from one year to the next.

facilitate efforts to implement similar school models at new campuses. We assess this quantitatively by studying variation in teacher value-added at charter and district schools. Teacher value-added estimates come from the following model for achievement of student i in grade g in calendar year t:

(5) 
$$Y_{igt} = \alpha_g + \lambda_t + X'_{igt}\gamma + \beta_{s(i,g)} + \theta_{j(i,g)t} + \delta_{c(i,g,t)} + \xi_i + \varepsilon_{igt}$$

The control vector  $X_{igt}$  includes student demographic characteristics and lagged test scores, as well as classroom-level averages of these variables. We also include grade  $(\alpha_g)$  and calendar year  $(\lambda_t)$  fixed effects. The function s(i,g) labels the school that student *i* attends for grade *g*, j(i,g) describes the identity of her grade *g* teacher, and c(i,g,t) denotes a specific classroom. Because classroom-level averages of the observables are included as controls, equation (5) describes a "correlated random effects" model in which the mean of the teacher effect distribution may depend on the characteristics of students in the classroom (Mundlak 1978, Chamberlain 1982). In other words, we are not imposing independence of teacher quality from student observables.

We also allow school and teacher effects to depend on observed school and teacher characteristics. The mean of the distribution of school effects  $\beta_s$  differs for charter and traditional public schools. The teacher effects (which measure variation in teacher effectiveness within school) are in turn written

$$\theta_{it} = \theta_i^0 + W'_{it}\theta^w,$$

where  $W_{jt}$  includes teacher j's experience as of year t in one of three experience groups (novice, one to four years of experience, and greater than five years of experience) as well as interactions of charter status with experience. Given the small number of charter teachers in the sample, we do not separate teachers at proven

	Math			English		
	Charter (1)	BPS (2)	p -value: Charter = BPS (3)	Charter (4)	BPS (5)	p -value: Charter = BPS (6)
Experience profile						
One to four years of experience	0.085 (0.024)	0.094 (0.036)	0.835	0.044 (0.023)	0.091 (0.034)	0.261
Five or more years of experience	$\begin{array}{c} 0.053 \\ (0.035) \end{array}$	$\begin{array}{c} 0.080 \\ (0.038) \end{array}$	0.600	$\begin{array}{c} 0.047 \\ (0.033) \end{array}$	0.096 (0.034)	0.307
Random effect parameters: SD						
School	0.135 (0.028)	$0.102 \\ (0.016)$	0.315	$0.091 \\ (0.021)$	$0.111 \\ (0.015)$	0.449
Teacher	0.118 (0.012)	0.185 (0.010)	0.000	0.104 (0.012)	0.178 (0.008)	0.000
Class	0.076 (0.008)	0.152 (0.005)	0.000	0.081 (0.009)	0.116 (0.005)	0.001
Observations		48,4	16		54,0	75

TABLE 9—TEACHER VALUE-ADDED ESTIMATES

*Notes:* This table shows teacher value-added across years of experience and variance in teacher value-added at the school, teacher, and classroom level. These results come from maximum likelihood estimation of a random effects model with normally distributed teacher, school, and classroom effects. The model controls for student demographics and lagged test scores as well as class averages of these variables. The mean of the school effect varies with charter status, and the mean of the teacher effect varies by experience and experience interacted with charter status.

providers, expansions, and other charters for the purposes of the value-added model, nor do we estimate experience premia for each year.<sup>23</sup> We model the school effects  $\beta_s$ , within-school teacher effects  $\theta_j^0$ , and classroom effects  $\delta_c$  as normally distributed conditional on  $X_{igt}$ , with variances that differ in charter and traditional public schools. The student random effect  $\xi_i$  and idiosyncratic error  $\varepsilon_{igt}$  are also modeled as normally distributed. Random effects specifications of this sort are common in the literature on teacher value-added, and previous studies have argued that such models generate estimates of teacher effectiveness that exhibit little selection bias (Kane, Rockoff, and Staiger 2008; Chetty, Friedman, and Rockoff 2014).<sup>24</sup>

As can be seen in Table 9, maximum likelihood estimation of model (5) reveals two notable patterns. The first is revealed by comparing variation in school, teacher, and class effects across the charter and traditional sectors. Both charter and district schools have similar variation in school-level effectiveness. At the teacher and classroom levels, we find less variation in effectiveness in the charter sector. In math, the standard deviation of the teacher random effect  $\theta_j^0$  is  $0.12\sigma$  compared to  $0.19\sigma$  in BPS, while the standard deviation of the class effects are similarly compressed for English

 $<sup>^{23}</sup>$ Data for the value-added model are from 2011–2014, the years in which it is possible to link students, teachers, and classrooms in our data.

<sup>&</sup>lt;sup>24</sup> Our findings are robust to other approaches of measuring value-added. We estimated alternative specifications using teacher random effects, teacher and school random effects, teacher fixed effects, and school and teacher fixed effects. We also estimated models excluding students who attended both charter and BPS during middle school, and used finer measures of teacher experience. These alternative approaches yielded similar patterns of results.



FIGURE 3. DISTRIBUTION OF TEACHER-SPECIFIC POSTERIOR MEAN PREDICTIONS OF VALUE-ADDED

*Note:* This figure plots the distribution of teacher-specific posterior mean predictions of value-added for charter and Boston Public School (BPS) teachers.

scores: the standard deviation of  $\theta_j^0$  is  $0.11\sigma$  in charters versus  $0.18\sigma$  in BPS, and the standard deviation of  $\delta_c$  is  $0.08\sigma$  versus  $0.12\sigma$  in BPS.<sup>25</sup>

Figure 3 displays distributions of posterior mean predictions of individual teacher value-added based on estimates of equation (5) separately for charter and traditional public schools. Note that the charter school teacher impacts are centered at the mean of the charter school effects.<sup>26</sup> These distributions are visibly less diffuse than that of their traditional public school counterparts, and appear to show a compressed distribution of effects rather than a truncated tail on either end.<sup>27</sup>

Overall, the evidence in Table 9 and Figure 3 suggests that the charter sector reduces variation in teacher effectiveness within schools, which may be due to charters' centralized management of teachers and standardized instructional practices.<sup>28</sup> Charter schools might also hire a population of teachers with less variation in practices. The reduction in variation at the classroom level (which is typically attributed to random events like construction noise on test day) suggests some of this variation is systematic and can be reduced through standardized practices as well.

A second pattern revealed by the value-added analysis is that returns to teacher experience seem less pronounced in charter schools than in BPS. Comparing

<sup>27</sup> This result is somewhat speculative due to the noisy charter estimates. Note in addition that Figure 3 analyzes the distribution of posterior means rather than the true underlying distribution of latent teacher effects.
 <sup>28</sup> Taylor (2018) and Jackson and Makarin (2018) also show compression of teacher value-added distribu-

<sup>26</sup>Taylor (2018) and Jackson and Makarin (2018) also show compression of teacher value-added distributions in settings with standardized instructional practices. In Taylor (2018), standardization comes from the use of computer-aided instruction; in Jackson and Makarin (2018), standardization comes from access to high-quality instructional materials.

<sup>&</sup>lt;sup>25</sup> The results found here—that charter value-added standard deviations are around 0.11 and district about 0.18—indicate that charter schools in Boston are toward the minimum known range of teacher value-added estimates, whereas Boston district schools are in the middle of the distribution. Hanushek and Rivkin (2010) review the dispersion of teacher value-added in 10 localities, and find that the standard deviation of teacher effects ranges between 0.11 and 0.36 $\sigma$  in math and 0.10 and 0.26 $\sigma$  in reading.

<sup>&</sup>lt;sup>26</sup>This implies that comparisons between teachers are made within but not between sectors (BPS and charter). Cross-sector comparisons would require an analysis based on teachers switches between BPS and charter schools, which occur infrequently in our sample.

teachers with one to four years of experience and teachers with five or more years of experience to novices shows that more experienced teachers generally outperform new teachers. However, the experience premium is larger in BPS (though the differences are not statistically significant), with teachers with one to four years of experience outperforming novice teachers in BPS by about  $0.09\sigma$  in both math and English. The corresponding experience premium for teachers in charter schools is similar in math but about half the size in English, at  $0.04\sigma$ . For teachers with more than five years of experience, BPS teachers maintain their edge relative to novices, but any premium for charter school teachers is small and not statistically significant (though again the experience profiles in charter schools and BPS are not statistically distinguishable). In short, either through selection of teachers or through training, charter schools appear to dampen one of the most persistent findings in the literature on teacher effectiveness (Harris and Sass 2011; Papay and Kraft 2015; Clotfelter, Ladd, and Vigdor 2007; Rockoff 2004)-that teachers make significant gains in their first few years of teaching. Teachers at charter schools deliver effective education despite the high proportion of novice teachers and substantial teacher turnover. Taken together, the conclusions from the value-added analysis are consistent with the hypothesis that highly standardized management practices may contribute to the successful replication of charter school effects.

#### VII. Conclusion

The replication and expansion of successful schools is one strategy to address persistent achievement gaps in the United States. The efficacy of this strategy requires schools selected for expansion to maintain their success at new locations and with new student populations. Previous research has shown that urban No Excuses charter schools boost test scores markedly for small groups of applicants, suggesting the potential for transformational effects on urban achievement if these gains can be maintained at larger scales. We examine a recent policy change in Massachusetts that doubled Boston's charter sector over a short time period, allowing us to evaluate changes in the effects of No Excuses charters as these schools expanded to serve a larger share of the population within a single school market.

Our results demonstrate that Boston's No Excuses charters reproduced their effectiveness at new campuses. Lottery-based estimates show that schools selected for expansion produced larger gains than other charters in the pre-reform period, indicating that Massachusetts' accountability regime successfully identified more successful schools. New expansion campuses generate test score gains similar to those of their parent campuses despite a doubling of charter market share in middle school.

The demographics of students served by expansion charters are similar to those of the Boston population as a whole, suggesting that charter effectiveness is not driven by unique peer environments. We find that changes in student populations and the quality of fallback traditional public schools play only a small role in the effectiveness of charter expansion. Both a qualitative analysis of organizational practices during expansion and a quantitative analysis of variation in teacher value-added indicate that charter schools use a highly standardized model that limits variation in practices across schools and classrooms. This standardized approach may facilitate the portability of charter effectiveness to new campuses. More broadly, the role of these and other organizational practices in explaining successful replication of social programs is an important area for future work.

This paper also provides evidence on the efficacy of different organizational forms for replicating social programs. When a program is successful, policymakers face the decision of whether to have the original implementer continue to provide the program, or whether governments or other agencies should take over the program at a larger scale. This paper shows that in the charter school context, replicating existing charters is a viable strategy for charter expansion. This is consistent with the findings of Bold et al. (2018), who show that the successful Kenyan contract teacher program evaluated in Duflo, Dupas, and Kremer (2011, 2015) was replicated with provision by the original provider but not by the government (despite an identical contract). The "proven provider" design of the Massachusetts 2010 charter law is unique among the states with charter school laws, and it remains to be seen if other states or charter authorizers adopt such policies.<sup>29</sup> However, the share of charter schools managed by charter school management organizations (independent, nonprofit organizations that manage two or more charter schools) has grown from 16 percent in 2009 (Furgeson et al. 2011) to 23 percent in 2017 (David 2018), indicating that the market may institute a replication strategy even if authorizers do not.

### REFERENCES

- Abadie, Alberto. 2002. "Bootstrap Tests for Distributional Treatment Effects in Instrumental Variable Models." *Journal of the American Statistical Association* 97 (457): 284–92.
- Abdulkadiroğlu, Atila, Joshua D. Angrist, Susan M. Dynarski, Thomas J. Kane, and Parag A. Pathak. 2011. "Accountability and Flexibility in Public Schools: Evidence from Boston's Charters and Pilots." *Quarterly Journal of Economics* 126 (2): 699–748.
- Abdulkadiroğlu, Atila, Joshua D. Angrist, Peter D. Hull, and Parag A. Pathak. 2016. "Charters without Lotteries: Testing Takeovers in New Orleans and Boston." *American Economic Review* 106 (7): 1878–1920.
- Abdulkadiroğlu, Atila, Joshua D. Angrist, Yusuke Narita, and Parag A. Pathak. 2017. "Research Design Meets Market Design: Using Centralized Assignment for Impact Evaluation." *Econometrica* 85 (5): 1373–1432.
- Angrist, Joshua D., Sarah R. Cohodes, Susan M. Dynarski, Parag A. Pathak, and Christopher R. Walters. 2016. "Stand and Deliver: Effects of Boston's Charter High Schools on College Preparation, Entry, and Choice." *Journal of Labor Economics* 34 (2): 275–318.
- Angrist, Joshua D., Susan M. Dynarski, Thomas J. Kane, Parag A. Pathak, and Christopher R. Walters. 2010. "Inputs and Impacts in Charter Schools: KIPP Lynn." *American Economic Review* 100 (2): 239–43.
- Angrist, Joshua D., Susan M. Dynarski, Thomas J. Kane, Parag A. Pathak, and Christopher R. Walters. 2012. "Who Benefits from KIPP?" *Journal of Policy Analysis and Management* 31 (4): 837–60.
- Angrist, Joshua D., Peter D. Hull, Parag A. Pathak, and Christopher R. Walters. 2017. "Leveraging Lotteries for School Value-Added: Testing and Estimation." *Quarterly Journal of Economics* 132 (2): 871–919.

<sup>29</sup> Denver Public Schools also has a separate process allowing both existing traditional public and charter schools to apply to open a replication campus. Similarly, the federal government supports charter replication through grant competitions; the Massachusetts law change is the only replication program we are aware of that has been codified into state law.

- Angrist, Joshua D., and Guido W. Imbens. 1995. "Two-Stage Least Squares Estimation of Average Causal Effects in Models with Variable Treatment Intensity." *Journal of the American Statistical Association* 90 (430): 431–42.
- Angrist, Joshua D., Parag A. Pathak, and Christopher R. Walters. 2013. "Explaining Charter School Effectiveness." American Economic Journal: Applied Economics 5 (4): 1–27.
- Arsen, David, and Yongmei Ni. 2011. "The Effects of Charter School Competition on School District Resource Allocation." *Educational Administration Quarterly* 48 (1): 3–38.
- Banerjee, Abhijit, Rukmini Banerji, James Berry, Esther Duflo, Harini Kannan, Shobhini Mukerji, Marc Shotland, and Michael Walton. 2017. "From Proof of Concept to Scalable Policies: Challenges and Solutions, with an Application." *Journal of Economic Perspectives* 31 (4): 73–102.
- Betts, Julian R. 2009. "The Competitive Effects of Charter Schools on Traditional Public Schools." In Handbook of Research on School Choice, edited by Mark Berends, Ann Primus and Matthew G. Springer, 195–208. New York: Routledge.
- **Bifulco, Robert, and Randall Reback.** 2014. "Fiscal Impacts of Charter Schools: Lessons from New York." *Education Finance and Policy* 9 (1): 86–107.
- Blinder, Alan S. 1973. "Wage Discrimination: Reduced form and Structural Estimates." *Journal of Human Resources* 8 (4): 436–55.
- **Bold, Tessa, Mwangi Kimenyi, Germano Mwabu, Alice Ng'ang'a, and Justin Sandefur.** 2018. "Experimental Evidence on Scaling up Education Reforms in Kenya." *Journal of Public Economics* 168: 1–20.
- Booker, Kevin, Scott M. Gilpatric, Timothy Gronberg, and Dennis Jansen. 2008. "The Effect of Charter Schools on Traditional Public School Students in Texas: Are Children Who Stay Behind Left Behind?" *Journal of Urban Economics* 64 (1): 123–45.
- Borman, Geoffrey D., Robert E. Slavin, Alan C.K. Cheung, Anne M. Chamberlain, Nancy A. Madden, and Bette Chambers. 2007. "Final Reading Outcomes of the National Randomized Field Trial of Success for All." *American Educational Research Journal* 44 (3): 701–31.
- **Boston Municipal Research Bureau.** 2008. *The Status of Charter Schools in Boston*. Boston: Boston Municipal Research Bureau.
- **Carter, Samuel Casey.** 2000. *No Excuses: Lessons from 21 High-Performing, High-Poverty Schools.* Washington, DC: The Heritage Foundation.
- Chabrier, Julia, Sarah Cohodes, and Philip Oreopoulos. 2016. "What Can We Learn from Charter School Lotteries." *Journal of Economic Perspectives* 30 (3): 57–84.
- **Chamberlain, Gary.** 1982. "Multivariate Regression Models for Panel Data." *Journal of Econometrics* 18 (1): 5–46.
- Chetty, Raj, John N. Friedman, and Jonah E. Rockoff. 2014. "Measuring the Impacts of Teachers I: Evaluating Bias in Teacher Value-Added Estimates." *American Economic Review* 104 (9): 2593–2632.
- Clotfelter, Charles T., Helen F. Ladd, and Jacob L. Vigdor. 2007. "How and Why Do Teacher Credentials Matter for Student Achievement?" NBER Working Paper 12828.
- Cohodes, Sarah R., Elizabeth M. Setren, and Christopher R. Walters. 2021. "Replication data for: Can Successful Schools Replicate? Scaling Up Boston's Charter School Sector." American Economic Association [publisher], Inter-university Consortium for Political and Social Research [distributor]. https://doi.org/10.3886/E118542V1.
- Cohodes, Sarah R., Elizabeth M. Setren, Christopher R. Walters, Joshua D. Angrist, and Parag A. Pathak. 2013. *Charter School Demand and Effectiveness: A Boston Update*. Boston, MA: The Boston Foundation and NewSchools Venture Fund.
- Cook, Jason B. 2018. "The Effect of Charter Competition on Unionized District Revenues and Resource Allocation." *Journal of Public Economics* 158: 48–62.
- **Cordes, Sarah A.** 2018. "In Pursuit of the Common Good: The Spillover Effects of Charter Schools on Public School Students in New York City." *Education Finance and Policy* 13 (4): 484–512.
- **David, Rebecca.** 2018. *National Charter School Management Overview: 2016-17 School Year.* Washington, DC: National Alliance for Public Charter Schools.
- David, Rebecca, Kevin Hesla, and Susan Aud Pendergrass. 2017. A Growing Movement: America's Largest Public Charter School Communities. Washington, DC: National Alliance for Public Charter Schools.
- Davis, Tomeka M. 2013. "Charter School Competition, Organization, and Achievement in Traditional Public Schools." *Education Policy Analysis Archives* 21 (88).
- Davis, Jonathan M.V., Jonathan Guryan, Kelly Hallberg, and Jens Ludwig. 2017. "The Economics of Scale-Up." NBER Working Paper 23925.

- Davis, Matthew, and Blake Heller. 2019. "No Excuses Charter Schools and College Enrollment: New Evidence from a High School Network in Chicago." *Education Finance and Policy* 14 (3): 414–40.
- de Chaisemartin, Clément, and Luc Behaghel. 2020. "Estimating the Effect of Treatments Allocated by Randomized Waiting Lists." *Econometrica* 88 (4): 1453–1477.
- **Dobbie, Will, and Roland G. Fryer, Jr.** 2011. "Are High-Quality Schools Enough to Increase Achievement among the Poor? Evidence from the Harlem Children's Zone." *American Economic Journal: Applied Economics* 3 (3): 158–87.
- **Dobbie, Will, and Roland G. Fryer, Jr.** 2013. "Getting beneath the Veil of Effective Schools: Evidence from New York City." *American Economic Journal: Applied Economics* 5 (4): 28–60.
- Dobbie, Will, and Roland G. Fryer, Jr. 2015. "The Medium-Term Impacts of High-Achieving Charter Schools." Journal of Political Economy 123 (5): 985–1037.
- **Duflo, Esther, Pascaline Dupas, and Michael Kremer.** 2011. "Peer Effects, Teacher Incentives, and the Impact of Tracking: Evidence from a Randomized Evaluation in Kenya." *American Economic Review* 101 (5): 1739–74.
- **Duflo, Esther, Pascaline Dupas, and Michael Kremer.** 2015. "School Governance, Teacher Incentives, and Pupil–Teacher Ratios: Experimental Evidence from Kenyan Primary Schools." *Journal of Public Economics* 123: 92–110.
- Epple, D., R. Romano, and R. Zimmer. 2016. "Charter Schools: A Survey of Research on Their Characteristics and Effectiveness." In *Handbook of the Economics of Education*, Vol. 5, edited by Eric A. Hanushek, Stephen Machin and Luger Woessmann, 139–208. North Holland: Elsevier.
- **Fryer, Roland G., Jr.** 2014. "Injecting Charter School Best Practices into Traditional Public Schools: Evidence from Field Experiments." *Quarterly Journal of Economics* 129 (3): 1355–1407.
- Furgeson, Joshua, Brian Gill, Joshua Haimson, Alexandra Killewald, Moira McCullough, Ira Nichols-Barrer, Bing-ru Teh, Natalya Verbitsky-Savitz, et al. 2011. The National Study of Charter Management Organization (CMO) Effectiveness: Charter-School Management Organizations: Diverse Strategies and Diverse Student Impacts. Princeton, NJ: Mathematica Policy Research.
- Gill, Brian P. 2016. "The Effect of Charter Schools on Students in Traditional Public Schools: A Review of the Evidence." *Education Next*, November 2. https://www.educationnext.org/the-effect-of-charter-schools-on-students-in-traditional-public-schools-a-review-of-the-evidence/.
- Gill, Brian, and Kevin Booker. 2008. "School Competition and Student Outcomes." In *Handbook of Research in Education Finance and Policy*, edited by Helen F. Ladd and Edward B. Fiske, 183–200. New York: Routledge.
- Hanushek, Eric A., and Steven G. Rivkin. 2010. "Generalizations about Using Value-Added Measures of Teacher Quality." *American Economic Review* 100 (2): 267–71.
- Harris, Douglas N., and Tim R. Sass. 2011. "Teacher Training, Teacher Quality and Student Achievement." *Journal of Public Economics* 95 (7–8): 798–812.
- Heckman, James, Seong Hyeok Moon, Rodrigo Pinto, Peter Savelyev, and Adam Yavitz. 2010. "Analyzing Social Experiments as Implemented: A Reexamination of the Evidence from the HighScope Perry Preschool Program." *Quantitative Economics* 1 (1): 1–46.
- Heckman, James, Rodrigo Pinto, and Peter Savelyev. 2013. "Understanding the Mechanisms through Which an Influential Early Childhood Program Boosted Adult Outcomes." *American Economic Review* 103 (6): 2052–86.
- Hoxby, Caroline M. 2003. "School Choice and School Productivity: Could School Choice Be a Tide that Lifts All Boats?" In *The Economics of School Choice*, edited by Caroline M. Hoxby, 287–342. Chicago: University of Chicago Press.
- **Imberman, Scott A.** 2011. "The Effect of Charter Schools on Achievement and Behavior of Public School Students." *Journal of Public Economics* 95 (7–8): 850–63.
- Jackson, C. Kirabo, and Elias Bruegmann. 2009. "Teaching Students and Teaching Each Other: The Importance of Peer Learning for Teachers." *American Economic Journal: Applied Economics* 1 (4): 85–108.
- Jackson, Kirabo, and Alexey Makarin. 2018. "Can Online Off-the-Shelf Lessons Improve Student Outcomes? Evidence from a Field Experiment." *American Economic Journal: Economic Policy* 10 (3): 226–54.
- Jepsen, Christopher, and Steven Rivkin. 2009. "Class Size Reduction and Student Achievement: The Potential Tradeoff between Teacher Quality and Class Size." *Journal of Human Resources* 44 (1): 223–50.
- Jinnai, Yusuke. 2014. "Direct and Indirect Impact of Charter Schools' Entry on Traditional Public Schools: New Evidence from North Carolina." *Economics Letters* 124 (3): 452–56.

- Kane, Thomas J., Jonah E. Rockoff, and Douglas O. Staiger. 2008. "What Does Certification Tell Us About Teacher Effectiveness? Evidence from New York City." *Economics of Education Review* 27 (6): 615–31.
- Krueger, Alan B. 1999. "Experimental Estimates of Education Production Functions." *Quarterly Journal of Economics* 114 (2): 497–532.
- Ladd, Helen F. 2002. "School Vouchers: A Critical View." *Journal of Economic Perspectives* 16 (4): 3–24.
- Ladd, Helen F., and John D. Singleton. 2020. "The Fiscal Externalities of Charter Schools: Evidence from North Carolina." *Education Finance and Policy* 15 (1): 191–208.
- Massachusetts Department of Elementary and Secondary Education. 2015. "Education Laws and Regulations 603 CMR 1.00: Charter Schools." Massachusetts Department of Elementary and Secondary Education. http://www.doe.mass.edu/lawsregs/603cmr1.html (accessed April 1, 2019).
- Massachusetts Department of Elementary and Secondary Education. 2016. "Massachusetts Charter Schools Fact Sheet, Director and Application History." http://www.doe.mass.edu/charter/factsheet. html (accessed April 1, 2019).
- Massachusetts State Legislature. 2010. An Act Relative to the Achievement Gap, Chap. 12. https://malegislature.gov/Laws/SessionLaws/Acts/2010/Chapter12.
- **Mundlak, Yair.** 1978. "On the Pooling of Time Series and Cross Section Data." *Econometrica* 46 (1): 69–85.
- **New York State Department of Education.** 2019. "New York Charter School Fact Sheet." http://www.p12.nysed.gov/psc/facts/nyscsfactsheet1042019.pdf (accessed April 1, 2019).
- Oaxaca, Ronald. 1973. "Male-Female Wage Differentials in Urban Labor Markets." *International Economic Review* 14 (3): 693–709.
- Papay, John P., and Matthew A. Kraft. 2015. "Productivity Returns to Experience in the Teacher Labor Market: Methodological Challenges and New Evidence on Long-Term Career Improvement." *Journal of Public Economics* 130: 105–19.
- Papay, John P., Eric S. Taylor, John H. Tyler, and Mary Laski. 2020. "Learning Job Skills from Colleagues at Work: Evidence from a Field Experiment Using Teacher Performance Data." American Economic Journal: Economic Policy 12 (1): 359–88.
- Puma, Mike, Stephen Bell, Ronna Cook, Camilla Heid, Pam Broene, Frank Jenkins, Andrew Mashburn, and Jason Downer. 2012. Third Grade Follow-up to the Head Start Impact Study. Washington, DC: Office of Planning, Research and Evaluation.
- Quint, Janet, Pei Zhu, Rekha Balu, Shelley Rappaport, and Micah DeLaurentis. 2015. Scaling Up the Success for All Model of School Reform. New York: Manpower Demonstration Research Corporation.
- **Ridley, Matthew, and Camille Terrier.** 2018. "Fiscal and Education Spillovers from Charter School Expansion." SEII Discussion Paper 2018.02.
- **Rockoff, Jonah E.** 2004. "The Impact of Individual Teachers on Student Achievement. Evidence from Panel Data." *American Economic Review* 94 (2): 247–52.
- Ronfeldt, Matthew, Susanna Owens Farmer, Kiel McQueen, and Jason A. Grissom. 2015. "Teacher Collaboration in Instructional Teams and Student Achievement." *American Educational Research Journal* 52 (3): 475–514.
- Rothstein, Richard. 2004. Class and Schools: Using Social, Economic, and Educational Reform to Close the Black-White Achievement Gap. Washington, DC: Economic Policy Institute.
- Sacerdote, Bruce. 2011. "Peer Effects in Education: How Might They Work, How Big Are They and How Much Do We Know Thus Far?" In *Handbook of the Economics of Education*, Vol. 3, edited by Eric A. Hanushek, Stephen Machin and Ludger Woessmann, 249–77. North Holland: Elsevier.
- Sass, Tim R. 2006. "Charter Schools and Student Achievement in Florida." *Education Finance and Policy* 1 (1): 91–122.
- Setren, Elizabeth. Forthcoming. "Targeted vs. General Education Investments: Evidence from Special Education and English Language Learners in Boston Charter Schools." *Journal of Human Resources*.
- Shin, Hyo Jeong, Bruce Fuller, and Luke Dauter. 2017. "Heterogeneous Effects of Charter Schools: Unpacking Family Selection and Achievement Growth in Los Angeles." *Journal of School Choice* 11 (1): 60–94.
- Sun, Min, Susanna Loeb, and Jason A. Grissom. 2017. "Building Teacher Teams: Evidence of Positive Spillovers from More Effective Colleagues." *Educational Evaluation and Policy Analysis* 39 (1): 104–25.

- Taylor, Eric S. 2018. "New Technology and Teacher Productivity." https://scholar.harvard.edu/files/ erictaylor/files/technology-teachers-jan-18.pdf.
- The New Teacher Project (TNTP). 2014. Greenhouse Schools in Boston: School Leadership Practices across a High-Performing Charter Sector. New York: TNTP.
- **Thernstrom, Abigail, and Stephen Thernstrom.** 2004. *No Excuses: Closing the Racial Gap in Learning.* New York: Simon and Schuster.
- Tuttle, Christina Clark, Brian Gill, Philip Gleason, Virginia Knechtel, Ira Nichols-Barrer, and Alexandra Resch. 2013. *KIPP Middle Schools: Impact on Achievement and Other Outcomes*. Washington, DC: Mathematica Policy Research.
- Tuttle, Christina Clark, Philip Gleason, Virginia Knechtel, Ira Nichols-Barrer, Kevin Booker, Gregory Chojnacki, Thomas Coen, and Lisbeth Goble. 2015. Understanding the Effect of KIPP as it Scales: Volume I, Impacts on Achievement and Other Outcomes. Washington, DC: Mathematica Policy Research.
- Walters, Christopher R. 2018. "The Demand for Effective Charter Schools." *Journal of Political Economy* 126 (6): 2179–2223.
- Winters, Marcus A. 2012. "Measuring the Effect of Charter Schools on Public School Student Achievement in an Urban Environment: Evidence from New York City." *Economics of Education Review* 31 (2): 293–301.
- Winters, Marcus A. 2020. Do Charter Schools Harm Traditional Public Schools? Years of Test-Score Data Suggest They Don't. New York, NY: Manhattan Institute.
- Ziebarth, Todd, and Louann Bierlein Palmer. 2018. Measuring Up to the Model: A Ranking of State Public Charter School Laws, 2018. Washington, DC: National Alliance for Public Charter Schools.
- Zimmer, Ron, and Richard Buddin. 2009. "Is Charter School Competition in California Improving the Performance of Traditional Public Schools?" *Public Administration Review* 69 (5): 831–45.
- Zimmer, Ron, Brian Gill, Kevin Booker, Stephane Lavertu, and Tim R. Sass, John Witte. 2009. Charter Schools in Eight States: Effects on Achievement, Attainment, Integration, and Competition. Santa Monica: Rand Corporation.

# Online Appendix

## Can Successful Schools Replicate? Scaling Up Boston's Charter School Sector

Sarah R. Cohodes, Elizabeth M. Setren, and Christopher R. Walters

### 1 Data Appendix

We use lottery records, student demographic and enrollment data, state standardized test scores, and school personnel files in this article. Lottery records collected from individual schools contain the list of applicants, offer status, and factors that affect an applicant's lottery odds, including sibling status, disqualifications, late applications, and applying from outside of Boston. The Student Information Management Systems (SIMS) dataset contains enrollment and demographic data for all public school students in Massachusetts. Student standardized test scores come from the state database for the Massachusetts Comprehensive Assessment System (MCAS). The Massachusetts Education Personnel Information Management Systems (EPIMS) database provides school staff information. Next we describe these datasets, the matching process, and sample construction.

### 1.1 Lottery records

Massachusetts legally requires charters to admit students via lottery when more students apply to a charter school than the number of available seats for a given grade. Our paper uses records from charter lotteries conducted between spring 2004 to spring 2013 for 14 charter schools accepting students in 5th or 6th grade. Each of the 14 schools contributes oversubscribed lottery data.<sup>1</sup> Schools vary in the grades they serve and in years of operation. Table A1 lists this information and the years each school contributes to the analysis. We exclude one school that did not provide lottery records (Smith Leadership Academy) and two schools that closed before the charter expansion (Uphams Corner Charter School in 2009 and Frederick Douglass Charter School in 2005). Lottery data typically includes applicants' names, dates of birth, and lottery and waitlist offer status. Offers to attend charter schools either occur on the day of the lottery (referred to as *immediate of fer*) or after the day of the lottery when students receive offers from the randomly sequenced waitlist as seats become available. In three out of the 65 lotteries in the study, the schools gave all applicants offers or did not give waitlist offers to non-siblings. Four lotteries did not distinguish the timing of the offers so we code the immediate offer variable to equal zero for these cohorts.

The Uncommon Schools/Roxbury Preparatory charter network held a single lottery for its three campuses in the Spring 2012 and Spring 2013 lotteries. When the school called a student's lottery number, the student could pick from the campuses that still had open seats. Our lottery records show which campus they picked at the time of the lottery. We find the last lottery number for each campus and code all students with better lottery numbers as having offers from that campus.

Uncommon Schools offered seats from the waitlist as they became available for individual campuses. Parents chose to accept or decline waitlist offers for single schools. If they declined, they were taken off the waitlist and would not be considered for seats at the other campuses.

### 1.2 Enrollment and demographics

The SIMS data contains individual level data for students enrolled in public schools in Massachusetts from 2003-2004 through 2013-2014. The data contains snapshots from October and the end of the school year. Each student has only one observation in each time period, except when

<sup>&</sup>lt;sup>1</sup>We do not have Spring 2004 lottery records for Brooke Roslindale, Boston Prep, and Academy of the Pacific Rim or Spring 2005 records for Brooke Roslindale. Brooke Roslindale does not have lotteries in after charter expansion because their elementary school students filled the middle school seat. All other schools and years have oversubscribed lottery data.

students switch grades or schools within year. Fields include a unique student identifier, grade level, year, name, date of birth, gender, ethnicity, special education status, limited English proficiency status, free or reduced price lunch status, school attended, suspensions, attendance rates, and days truant.

We code students as charter attendees in a school year if they attended a charter at any point during a year. Students who attend more than one charter school in a year are assigned to the charter they attended the longest. Students who attend more than one traditional public school and no charter schools in a year are assigned to the school they attended the longest. We randomly choose between schools if students have attendance ties between the most attended schools.

### 1.3 Test scores

This paper uses individual student math and English Language Arts (ELA) Massachusetts Comprehensive Assessment System (MCAS) test scores from 2003-2004 through 2013-2014. Massachusetts public school students take the exam each year in grades grades 5 through 8. Data includes the unique student identifier. We standardize the raw scores to to have a mean of zero within subjectgrade-year in Boston.

### 1.4 Staff records

The Education Personnel Information Management Systems (EPIMS) contains yearly staff level data for all employees in Massachusetts public schools. We use data collected in October of the 2007-08 through the 2013-14 school years. Data includes job position, school, full time equivalency, date of birth, date of hire for first public school job in Massachusetts, license status, and highly qualified status. We use the full time equivalency of all staff and teachers. If one school has two half time teachers, they are counted as one full time equivalent teacher. A teacher who teaches at multiple schools counts towards the staff statistics at each school.

### 1.5 Matching data

We use applicants' names, date of birth, grade, and year to match their lottery records to the state enrollment data. The applicants who uniquely and exactly match the grade, year, name, and date of birth (if available) in the state records are assigned to the matched unique student id. After this initial match, we strip names in the lottery and enrollment data of spaces, surnames, hyphens, and apostrophes. Unique matches after this cleaning are assigned to the matched unique student id. Then, we use reclink, a fuzzy matching STATA program, to suggest potential matches for the remaining students. This matches students with slight spelling differences and those who appear in one grade older or younger than the charter application grade. We hand check these suggested matches for accuracy. We search for the remaining unmatched students by hand in the data. Typically this last group contains name truncations, name misspellings, or first and last names in the wrong field.

The matching process assigns 95 percent of applicants to the state administrative records (see Table A3). Students who do not match either enroll in private, parochial, or out-of-state schools, have names and birthdates too common to match, or have spelling errors too extreme to match with confidence. Receiving a charter offer makes students 3.8 more likely to match to the data,

as shown in Table A3. As a result, our findings show causal estimates for the set of students who enroll in Massachusetts Public Schools.

We match the enrollment and demographic data to the student test scores using the unique student identifier. Students who attend out of state, private, or parochial schools do not have test score outcomes for their years outside of Massachusetts public schools.

### **1.6** Sample restrictions

We exclude applicants who receive higher or lower preference in the lottery. Late applicants, those who apply to the wrong grade, out-of-area applicants, and siblings fall into these categories and typically have no variation in offer status. When students have duplicate applications within an individual school's lottery, we keep only one application. If students apply to charter schools in different years, we use only the first application year. We restrict the sample to students with baseline demographics data, excluding students applying from outside of Massachusetts public schools. With these restrictions imposed, the original raw sample of applications narrows from 20,981 to 8,473.

### References

Massachusetts Department of Elementary and Secondary Education. 2011-2014. Educator Personnel Information Management System (EPIMS). Commonwealth of Massachusetts. Multiple electronic files.

Massachusetts Department of Elementary and Secondary Education. 2001-2014. Massachusetts Comprehensive Assessment System (MCAS). Commonwealth of Massachusetts. Multiple electronic files.

Massachusetts Department of Elementary and Secondary Education. 2011-2014. Student Course System (SCS). Commonwealth of Massachusetts. Multiple electronic files.

Massachusetts Department of Elementary and Secondary Education. 2001-2014. Student Information Management System (SIMS). Commonwealth of Massachusetts. Multiple electronic files.

### 2 Additional Tables

			Outcome Years In
	Year Opened	Grades	Analysis
	(1)	(2)	(3)
Proven Providers			
Roxbury Preparatory: Mission Hill Campus	1999 - 2000	5 - 8 (12)	2004-05 - 2013-14
Brooke Roslindale	2002 - 03	5 - 8	2006-07 - 2009-10
Excel East Boston	2003 - 04	5 - 9 (12)	2008-09 - 2013-14
MATCH Middle School	2008 - 09	6 - 8	2008-09 - 2013-14
Expansion Charters			
Roxbury Preparatory: Lucy Stone Campus	2011 - 12	5 - 8	2011-12 - 2013-14
Roxbury Preparatory: Dorchester Campus	2012 - 13	5 - 7 (8)	2012-13 - 2013-14
Brooke Mattapan	2011 - 12	5 - 8	2011-12 - 2013-14
Brooke East Boston	2012 - 13	5 - 7 (8)	2012-13 - 2013-14
Excel Orient Heights	2012 - 13	5 - 7 (8)	2012-13 - 2013-14
KIPP	2012 - 13	5 - 7 (8)	2012-13 - 2013-14
UP Academy Boston	2011 - 12	6 - 8	2011-12 - 2013-14
Other Charters			
Academy of the Pacific Rim	1997 - 98	5 - 12	2005-06 - 2013-14
Boston Collegiate	1998 - 99	5 - 12	2004-05 - 2013-14
Boston Prep	2004 - 05	6 - 12	2005-06 - 2013-14
Not Included in Study			
Helen Davis Leadership Academy	2003 - 04	6 - 8	declined to participate
Frederick Douglas Charter	2000 - 01	6 - 10	closed in 2004-05
Uphams Corner Charter	2002 - 03	5 - 8	closed in 2008-09

Table A1: Charter Middle Schools in Boston

Notes: This table lists Boston middle school charter schools by school type, opening year, grade levels, and outcome years included in the analysis. Grade levels shown in parentheses indicate planned enrollment grades which were not present at the time of analysis.

Table A2: Lottery Records

				5							
Year of application	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	All
Total number of records	341	739	913	1143	1422	1595	1467	4283	4312	4766	20981
Excluding disqualifed applications	341	738	911	1135	1404	1594	1444	4273	4305	4760	20905
Excluding late applications	340	738	909	1135	1363	1566	1397	4163	4196	4583	20390
Excluding out of area applications	340	733	900	1123	1353	1548	1379	4094	4071	4513	20054
Excluding siblings	300	677	836	1021	1223	1408	1249	3758	3760	4320	18552
Excluding records not matched to SIMS	266	634	801	1000	1181	1378	1179	3627	3573	4016	17655
Keep only first year of charter application	266	617	770	962	1093	1282	1038	3308	2962	3469	15767
Excluding repeat applications	266	617	770	962	1093	1282	1038	3308	2962	3458	15756
Reshaping to one record per student	265	523	586	760	868	963	812	2055	1715	1900	10447
Has baseline demographics and in Boston at baseline	176	382	437	571	679	722	623	1790	1499	1594	8473

Notes: This table summarizes the sample restrictions imposed for the lottery analysis. Disqualified applications are duplicate records and applications to the wrong grade.

			Regression of Match on Offer			
	Number of	Proportion				
	Applications	Matched	Immediate Offer	Any Offer		
Lottery Year	(1)	(2)	(3)	(4)		
2004	268	0.989	-0.006	-0.007		
			(0.026)	(0.013)		
2005	616	0.987	-	0.002		
			-	(0.013)		
2006	742	0.991	-	0.004		
			-	(0.016)		
2007	924	0.984	0.019**	0.034***		
			(0.008)	(0.013)		
2008	1018	0.957	0.042***	0.061***		
			(0.013)	(0.019)		
2009	1106	0.977	0.004	0.011		
			(0.011)	(0.010)		
2010	1041	0.924	0.065***	0.071***		
			(0.016)	(0.017)		
2011	2614	0.954	0.018***	0.025***		
			(0.007)	(0.007)		
2012	2503	0.939	0.001	0.033***		
			(0.011)	(0.011)		
2013	2712	0.902	0.045***	0.078***		
			(0.012)	(0.015)		
All Cohorts	15482	0.949	0.023***	0.038***		
			(0.003)	(0.004)		

Notes: This table summarizes the match from the lottery records to administrative student data. The sample excludes late applicants, siblings, disqualified applicants, duplicate names, and out-of-area applicants. Columns (3) and (4) report coefficients from regressions on a dummy for a successful match on immediate and any charter offer dummies. All regressions control for school-by-year dummies. \*significant at 10%; \*\*significant at 5%; \*\*\*significant at 1%

		Befor		After Charter Expansion				
			2	SLS		2SLS		
		Non-Charter	Proven	Other Charters	Non-Charter	Proven	Expansion	Other
		Mean	Providers		Mean	Providers	Charters	Charters
		(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Math	0.117	0.337***	0.201***	-0.074	0.314***	0.218**	0.207***
~			(0.043)	(0.025)		(0.087)	(0.092)	(0.044)
ω	N	I			17395			
	English	0.201	0.162***	0.091***	-0.032	0.155*	0.202**	0.105**
			(0.043)	(0.024)		(0.089)	(0.094)	(0.044)
	N	I			17316			

### Table A4: Alternative Definition of Proven Provider & Replicate

Notes: This table reports the main 2SLS results from Table 6, but using alternative charter school type categorizations. In this robustness check, MATCH Middle School, UP Academy Boston, and KIPP Boston are considered other charters. \*significant at 10%; \*\*significant at 5%; \*\*\*significant at 1%

	Before Chart	er Expansion	After	r Charter Exp	ansion
	Proven	Other	Proven	Expansion	
	Providers	Charters	Providers	Schools	Other Charters
	(1)	(2)	(3)	(4)	(5)
Female	0.000	-0.004	-0.005	0.011	0.020
	(0.034)	(0.028)	(0.027)	(0.027)	(0.028)
Black	-0.026	0.007	-0.027	-0.025	-0.015
	(0.032)	(0.027)	(0.027)	(0.026)	(0.028)
Latino/a	0.027	0.000	-0.001	0.005	-0.010
	(0.031)	(0.022)	(0.027)	(0.026)	(0.027)
Asian	-0.014	0.007	0.008	0.010	0.000
	(0.009)	(0.008)	(0.010)	(0.011)	(0.009)
White	0.016	-0.003	0.007	0.001	0.018
	(0.011)	(0.024)	(0.010)	(0.012)	(0.017)
Subsidized Lunch	0.015	0.010	-0.011	-0.016	-0.016
	(0.029)	(0.027)	(0.020)	(0.019)	(0.023)
English Language Learners	-0.005	-0.001	-0.004	-0.039	-0.027
	(0.023)	(0.014)	(0.027)	(0.026)	(0.025)
Special Education	-0.005	0.005	0.002	0.013	0.018
	(0.027)	(0.022)	(0.021)	(0.022)	(0.022)
Attended charter before applying	0.010	-0.008	-0.015	-0.015*	-0.003
	(0.019)	(0.020)	(0.010)	(0.008)	(0.014)
Baseline math score	-0.024	-0.022	0.058	-0.033	-0.004
	(0.071)	(0.053)	(0.050)	(0.051)	(0.055)
Baseline English score	-0.036	0.000	0.048	0.037	0.011
	(0.071)	(0.052)	(0.053)	(0.051)	(0.055)
N (offered)	1009	1309	1466	1825	1142
P-value	0.594	0.891	0.526	0.136	0.979

### Table A5: Covariate Balance

Notes: This table reports coefficients from regressions of baseline characteristics on charter offers, controlling for lottery risk set indicators. P-values are from tests of the hypothesis that all coefficients are zero. \*significant at 10%; \*\*significant at 5%; \*\*\*significant at 1%

	זמטוב הט, הננוזנוטוו											
	Before <b>G</b>	Charter Expans	ion	A	After Charter Expansion							
		Offer Dif	ferential		0	Offer Differential						
	Non-offered	Proven	Other	Non-offered	Proven	Expansion	Other					
	Followup Rate	Providers	Charters	Followup Rate	Providers	Charters	Charters					
	(1)	(2)	(3)	(4)	(5)	(6)	(7)					
Math	0.834	0.018	0.032**	0.869	0.000	0.013	-0.023					
		(0.018)	(0.015)		(0.015)	(0.016)	(0.018)					
Ν	1			20102								
English	0.825	0.018	0.034**	0.869	0.001	0.011	-0.025					
		(0.017)	(0.015)		(0.015)	(0.016)	(0.018)					
Ν	J			20102								

Table A6: Attrition

Notes: This table investigates attrition for randomized charter school lottery applicants. Columns (1) and (4) report fractions of follow-up test scores in grades five through eight that are observed for students not offered seats. Columns (2)-(3) and (5)-(7) report coefficients from regressions of a follow-up indicator on a lottery offer indicator (immediate or waitlist) and students not offered seats. Regressions control for lottery risk sets, as well as gender, ethnicity, a female-minority interaction, special education, English language learner, subsidized lunch status, and grade and year indicators. Standard errors are clustered by student.

Table A7: Overall Charter Effects									
	First Stage Reduced Form 2SLS								
	(1)	(2)	(3)						
Math	0.978***	0.218***	0.223***						
	(0.025)	(0.023)	(0.023)						
	N	17395							
English	0.977***	0.120***	0.123***						
	(0.025)	(0.022)	(0.022)						
	N	17316							

Notes: This table reports first stage, reduced form, and 2SLS estimates for the full sample of lotteried charter middle schools across all years and schools. The endogenous variable is years in any charter school and the instrument is any charter offer. The sample stacks post-lottery test scores in grades five through eight. Models control for baseline covariates and lottery risk sets. Standard errors are clustered by student.

	Before Chart	er Expansion	Afte	After Charter Expansion			
	Proven	Other	Proven	Expansion	Other		
	Providers (1)	Charters (2)	Providers (3)	Charters (4)	Charters (5)		
Math	0.386***	0.146***	0.157***	0.119**	0.102**		
	(0.047)	(0.039)	(0.046)	(0.046)	(0.051)		
English	0.157***	0.095***	0.069	0.093**	0.056		
-	(0.044)	(0.036)	(0.043)	(0.045)	(0.048)		

Table A8: Reduced Form Charter Effects on Test Scores

Notes: This table shows reduced form estimates of the effects of charter offers on math and English scores. Charter offer equals one if the student receives an immediate or a waitlist offer. See Table 5 for detailed regression specification notes.

	Before Charter Expansion				After Charter Expansion				
		O	LS			OLS			
	Non-Charter	Proven	Other	Non-Charter	Proven	Expansion	Other		
	Mean	Providers	Charters	Mean	Providers	Charters	Charters		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)		
Math	0.004	0.355***	0.228***	-0.059	0.299***	0.322***	0.212***		
		(0.010)	(0.010)		(0.011)	(0.014)	(0.010)		
N (students) N (scores)				31218 84246					
English	0.009	0.268*** (0.010)	0.088*** (0.011)	-0.032	0.185*** (0.012)	0.161*** (0.013)	0.181*** (0.010)		
N (students)				31242					
N (scores)				84290					

Table A9: Ordinary Least Squares Estimates of Char	ter Effects
--	-------------

Notes: This table reports ordinary least squares (OLS) estimates of the effects of time spent in charter schools on math and English scores for students who attend a school in Boston in the fourth grade. The sample stacks scores in grades five through eight for all Boston students. All regressions control for fourth grade math and English scores, as well as gender, ethnicity, a female-minority interaction, special education, English language learner, subsidized lunch status and grade and year indicators.

	Befor	e Charter Expar	sion		After Charter Expansion			
		Estim	nates			Estimates		
	Non-Charter	Proven	Other	Non-Charter	Proven	Expansion	Other	
	Mean	Providers	Charters	Mean	Providers	Charters	Charters	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
		Panel A: Fi	rst Stage Estimat	es				
Immediate Offer		0.748***	0.768***		0.572***	0.431***	0.620***	
		(0.031)	(0.021)		(0.031)	(0.028)	(0.031)	
Waitlist Offer		0.585***	0.471***		0.274***	0.258***	0.491***	
		(0.025)	(0.029)		(0.026)	(0.026)	(0.039)	
		<u>Panel B:</u>	2SLS Estimates					
Math	0.112	0.482***	0.201***	-0.054	0.521***	0.305**	0.076	
		(0.075)	(0.063)		(0.106)	(0.121)	(0.086)	
<i>P</i> -value: Equals proven provider <i>P</i> -value: Equals other charters			0.001			0.081 0.058	0.000	
N (Applicants)	1172	1245	1769	2694	2281	2387	2375	
N (Total scores)			7	087				
English	0.221	0.095	-0.045	-0.051	0.231**	0.284**	0.034	
		(0.078)	(0.061)		(0.112)	(0.129)	(0.089)	
<i>P</i> -value: Equals proven provider			0.112			0.698	0.121	
<i>P</i> -value: Equals other charters						0.045		
N (Applicants)	1138	1159	1762	2697	2286	2395	2382	
N (Total scores)			7	006				

Table A10: Charter Effects on Test Scores One Year After the Lotter	ry
---	----

Notes: Panel A reports the first stage effects of charter lottery offers on enrollment in a charter school in the first year after the lottery. Panel B displays the 2SLS estimates of the effects of charter school enrollment in the first year after the lottery on test scores. The endogenous variables are indicators of charter enrollment for the different charter types (pre-expansion proven providers, pre-expansion other charters, post-expansion proven providers, expansion schools, and post-expansion other charters). The instruments are immediate and waitlist lottery offer dummies for each school type. Immediate offer equals one for applicants offered seats on the day of the lottery. Waitlist offer equals one for applicants offered seats from the waitlist. Controls include lottery risk sets, as well as gender, race, ethnicity, a female-minority interaction, special education, English language learner, subsidized lunch status, and grade and year indicators. \*significant at 10%; \*\*significant at 5%; \*\*\*significant at 1%

	Before Charter Expansion			After Charter Expansion				
	Estimates							
	Non-Charter	Proven	Other	Non-Charter	Proven	Expansion	Other	
	Mean	Providers	Charters	Mean	Providers	Charters	Charters	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
		<u>Panel A: Fi</u>	rst Stage Estimat	es				
Immediate Offer		0.855***	1.185***		0.710***	0.484***	0.854***	
		(0.070)	(0.048)		(0.054)	(0.045)	(0.054)	
		Panel B:	2SLS Estimates					
Math	0.117	0.294***	0.201***	-0.074	0.412***	0.408***	0.218***	
		(0.061)	(0.031)		(0.082)	(0.085)	(0.068)	
<i>P</i> -value: Equals proven provider			0.108			0.960	0.041	
<i>P</i> -value: Equals other charters						0.036		
N (Applicants)	1093	1279	1909	2443	2303	2416	2405	
N (Total scores)			12	7395				
English	0.201	0.165***	0.083***	-0.032	0.222**	0.267***	0.151**	
		(0.061)	(0.030)		(0.090)	(0.086)	(0.067)	
P -value: Equals proven provider			0.153			0.638	0.479	
<i>P</i> -value: Equals other charters						0.197		
N (Applicants)	1087	1277	1911	2441	2307	2420	2412	
N (Total scores)			12	7316				

Table A11: Charter	Effects on	<b>Test Scores</b>	with Immediate	Offer	Instruments	Only	ÿ
							/

Notes: Panel A reports the first stage effects of charter lottery offers on years of enrollment in charter schools. Panel B displays the 2SLS estimates of the effects of charter school attendance on test scores. The sample stacks post-lottery test scores in grades five through eight. The endogenous variables are counts of years spent in the different charter types (pre-expansion proven providers, pre-expansion other charters, post-expansion proven providers, expansion schools, and post-expansion other charters). The instruments are immediate offer dummies for each school type. Immediate offer equals one for applicants offered seats on the day of the lottery. Controls include lottery risk sets, as well as gender, race, ethnicity, a female-minority interaction, special education, English language learner, subsidized lunch status, and grade and year indicators. Standard errors are clustered by student.

	Before Charter Expansion			After Charter Expansion				
	Non-Charter	2SLS		Non-Charter		2SLS		
	Mean	Proven	Other	Mean	Proven	Expansion	Other	
		Providers	Charters		Providers	Charters	Charters	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
	<u>Panel</u>	A: Peer Ouality: I	Peer Baseline Sum	of Math and English	<u>h</u>			
First Post-lotto Year	-0.979	0.234***	0.405***	-1.016	0.212***	0.174*	0.334***	
		(0.059)	(0.042)		(0.070)	(0.090)	(0.059)	
N (applicants)	1163	1102	1728	2803	2360	2474	2456	
N (total)				70	89			
Second Post-lotto Year	-0.824	0.168***	0.148***	-0.989	0.221***	0.102	0.197***	
		(0.059)	(0.039)		(0.064)	(0.084)	(0.057)	
N (applicants)	960	1050	1648	2295	2223	2327	2311	
N (total)				67.	53			
Third Post-lotto Year	-0.699	0.078*	0.054*	-0.944	0.031	0.023	0.060	
		(0.044)	(0.031)		(0.059)	(0.079)	(0.052)	
N (applicants)	926	1043	1644	1486	1375	1560	1355	
N (total)				512	23			
		Panel B: Schoo	ol Switching Afte	r One Yea <u>r</u>				
Any Switch	0.499	-0.257***	-0.521***	0.500	-0.337***	-0.263***	-0.489***	
		(0.047)	(0.043)		(0.077)	(0.096)	(0.072)	
Switch Excluding Transition Grades	0.176	-0.064*	-0.140***	0.178	-0.057	-0.072	-0.088*	
		(0.035)	(0.031)		(0.056)	(0.070)	(0.053)	
Ever Attend Exam School	0.082	-0.053**	-0.033*	0.053	-0.021	-0.014	0.013	
		(0.021)	(0.019)		(0.034)	(0.043)	(0.032)	

### A12: Lottery Estimates of Effects on Peer Quality and School Switching

Notes: This table displays 2SLS effects of charter enrollment in different types of charter schools on peer quality and switching schools on year after the lottery. Peer quality is defined as the average baseline test score math and English total for the other students in the student's school and grade. Students who do not appear in Massachusetts public schools in October following the charter application are not counted as school switchers. The switch excluding transitional grades equals one for students who switch schools in grades other than the exit grade of their first school. It does not equal one if the school closed the year the student switched. See Table 5 for detailed regression specification notes.

_	Math scores					English scores				
-	Before expansion Afte		fter expansio	er expansion		Before expansion		After expansion		
	Proven	Other	Proven	Expansion	Other	Proven	Other	Proven	Expansion	Other
	Providers	Charters	Providers	Charters	Charters	Providers	Charters	Providers	Charters	Charters
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
English Language	0.288***	-0.194	0.505***	0.284*	0.329***	0.165	-0.254*	0.334***	0.220	0.233*
Learner	(0.088)	(0.157)	(0.101)	(0.148)	(0.118)	(0.101)	(0.138)	(0.108)	(0.146)	(0.120)
N (scores)	468	455	1729	1804	1275	468	454	1733	1807	1279
Not English	0.330***	0.192***	0.252***	0.334***	0.145**	0.127***	0.097***	0.090	0.242***	0.092
Language Learner	(0.040)	(0.027)	(0.093)	(0.082)	(0.067)	(0.038)	(0.025)	(0.097)	(0.084)	(0.064)
N (scores)	3368	5640	2567	2955	3077	3286	5630	2565	2962	3084
Special Education	0.217**	0.156**	0.242	0.628***	0.180	0.039	0.119*	0.130	0.301	0.165
	(0.103)	(0.064)	(0.189)	(0.177)	(0.212)	(0.117)	(0.062)	(0.204)	(0.202)	(0.227)
N (scores)	693	1178	823	930	758	683	1171	818	936	763
Not Special	0.346***	0.184***	0.407***	0.270***	0.190***	0.158***	0.091***	0.232***	0.221***	0.109*
Education	(0.039)	(0.029)	(0.073)	(0.082)	(0.060)	(0.036)	(0.027)	(0.076)	(0.080)	(0.058)
	5145	4917	5475	3829	3394	3071	4913	5460	3833	3000
Below-mean	0.356***	0.239***	0.484***	0.481***	0.181**	0.148**	0.112**	0.324***	0.321***	0.165*
baseline score	(0.057)	(0.043)	(0.094)	(0.104)	(0.073)	(0.068)	(0.050)	(0.108)	(0.098)	(0.087)
N (scores)	1488	2072	2150	2265	1901	1320	1865	1964	2211	1727
Above-mean	0.343***	0.161***	0.207***	0.280***	0.240***	0.181***	0.080***	0.017	0.180**	0.132**
baseline score	(0.035)	(0.026)	(0.079)	(0.072)	(0.057)	(0.032)	(0.023)	(0.083)	(0.077)	(0.060)
N (scores)	2348	4023	2146	2494	2451	2434	4219	2334	2558	2636

Table A13: Charter School Effects for Subgroups

Notes: This table reports 2SLS estimates of the effects of charter school attendance on test scores for subgroups of students. The sample stacks post-lottery test scores in grades five through eight. The endogenous variables are counts of years spent in the different charter types. The instruments are immediate and waitlist lottery offer dummies for each school type. Controls include lottery risk sets, as well as gender, ethnicity, a female-minority interaction, special education, English language learner, subsidized lunch status, and grade and year indicators.